



Mobile Phone Data: A Survey of Techniques, Features, and Applications

Mohammed Okmi^{1,2}, Lip Yee Por^{1,*}, Tan Fong Ang¹ and Chin Soon Ku^{3,*}

- ¹ Faculty of Computer Science and Information Technology, University of Malaya, Kuala Lumpur 50603, Malaysia
- ² Department of Information Technology and Security, Jazan University, Jazan 45142, Saudi Arabia
- ³ Department of Computer Science, Universiti Tunku Abdul Rahman, Kampar 31900, Malaysia
- * Correspondence: porlip@um.edu.my (L.Y.P.); kucs@utar.edu.my (C.S.K.)

Abstract: Due to the rapid growth in the use of smartphones, the digital traces (e.g., mobile phone data, call detail records) left by the use of these devices have been widely employed to assess and predict human communication behaviors and mobility patterns in various disciplines and domains, such as urban sensing, epidemiology, public transportation, data protection, and criminology. These digital traces provide significant spatiotemporal (geospatial and time-related) data, revealing people's mobility patterns as well as communication (incoming and outgoing calls) data, revealing people's social networks and interactions. Thus, service providers collect smartphone data by recording the details of every user activity or interaction (e.g., making a phone call, sending a text message, or accessing the internet) done using a smartphone and storing these details on their databases. This paper surveys different methods and approaches for assessing and predicting human communication behaviors and mobility patterns from mobile phone data and differentiates them in terms of their strengths and weaknesses. It also gives information about spatial, temporal, and call characteristics that have been extracted from mobile phone data and used to model how people communicate and move. We survey mobile phone data research published between 2013 and 2021 from eight main databases, namely, the ACM Digital Library, IEEE Xplore, MDPI, SAGE, Science Direct, Scopus, SpringerLink, and Web of Science. Based on our inclusion and exclusion criteria, 148 studies were selected.



Citation: Okmi, M.; Por, L.Y.; Ang, T.F.; Ku, C.S. Mobile Phone Data: A Survey of Techniques, Features, and Applications. *Sensors* **2023**, *23*, 908. https://doi.org/10.3390/s23020908

Academic Editor: Giovanni Pau

Received: 21 November 2022 Revised: 22 December 2022 Accepted: 29 December 2022 Published: 12 January 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). **Keywords:** mobile phone data; call detail records (CDRs); mobility patterns; communication behaviors; urban crime patterns; urban sensors; smartphones

1. Introduction

In the past few decades, mobile phone data has grown into a stand-alone topic [1]. Due to the wide use of smartphones [2], the digital traces left by smartphone use have come to provide valuable real-time information about human activities. These digital traces facilitate the study of human behaviors. Various techniques and analytical perspectives have been employed to capture many aspects of human behaviors from mobile phone data, which has resulted in various applications. These analytical perspectives, such as human mobility, communication patterns, social networks, and mobile phone usage activities, have been built on various spatiotemporal and call characteristics extracted from mobile phone data.

For example, in the context of criminology, refs. [3,4] constructed a criminal network based on the communication traces that had been left by criminals over a period of time. The traces included details about the locations where the criminals had received calls, the timestamps of their communications, the frequency of their calls, and the periods of their activity. When these traces were collected over a period of time and put together, they provided insights into the communication patterns of the individuals concerned,

which were then used to infer their criminal behavior and relationships. Griffiths et al. [5] analyzed the mobility traces of criminals based on the digital traces they had left at their homes and at other meaningful locations (e.g., the crime scene) to determine whether the criminals' movements were regular or random. The researchers subsequently concluded that there was a high degree of spatial regularity in the criminals' movements. In the field of urban sensing, the same process has been used for a number of other things, such as figuring out people's social ties and interactions [6] and figuring out how the population changes every day [7].

Spatiotemporal mobility patterns and mobile phone activity patterns have also been widely extracted from mobile phone data for other practical applications, such as capturing individuals' activities associated with urban zones, transportation, and the COVID-19 pandemic. The authors of [8-12], among others, explored or captured human activity patterns from mobile phone data to infer land use types based on spatiotemporal call volume patterns. This feature represents the total call volume (the number of incoming calls received, and outgoing calls made from all smartphones) managed by a given base transceiver station (BTS) within a given period of time. On the other hand, Mao et al. [13], Lenormand et al. [14], and Ríos and Muñoz [15] inferred land use based on temporal changes in human activities. For example, Ríos and Muñoz [15] detected land use patterns based on changes over time (temporal changes) in human activities. They did this by extracting the spatiotemporal features from mobile phone data, such as the number of calls managed by every BTS every hour per week. Empirically, to identify human behavior patterns based on temporal changes, the activity pattern (total number of calls) of each BTS in a week (from Sunday to Saturday and every hour) is depicted by the number of calls managed by the BTS every hour over a seven-day week. This results in a list of consecutive time frames (time windows) [15] or creates a 7-day/24-h call pattern time series.

Another example of practical application of extracting mobile phone activity patterns from mobile phone data is mapping population distributions based on mobility patterns that shape human presence and mobility in space and time [16] and allow for understanding the spatiotemporal dynamics of an urban population [17]. Here, multiple spatiotemporal features have been extracted to depict human mobility patterns. For example, the cell tower ID shows the geographical location of the smartphone from which a call was made or received (smartphones connect regularly to the nearest cell towers), along with a timestamp record of when the interactive event happened. It also shows whether users are more active during the daytime or nighttime [17], based on when the calls were made.

We noticed that the human activity characteristics extracted from mobile phone data varied from one study to another. The concept of circadian rhythm in human activities has been empirically depicted in different ways. Thus, as mentioned earlier, this survey was conducted to provide a detailed description of the multiple features extracted from mobile phone data to depict human activities. Furthermore, the extant literature provides various processing techniques and analytical perspectives to capture and depict different aspects of human behavior, such as human mobility patterns, social networks, and communication behaviors. As mentioned earlier, these analytical perspectives have been built based on multiple spatiotemporal and call characteristics and features extracted from mobile phone data. Additionally, different types of mobile phone data show that different spatiotemporal and call characteristics can be extracted to map people's behaviors. The goal of this survey was to figure out what these characteristics are and what they do.

In short, this survey looks at the aspects of human behavior (mobility, communication, and social networks) that can be learned about by analyzing mobile phone data. This includes the spatiotemporal and call features of the data, which show patterns in how people move and talk to each other, as well as the methods that have been used to analyze mobile phone data to learn about these things.

The primary objective of this review is to investigate the current state of mobile phone data applications in various domains, with a specific focus on topics that the current literature has not covered, especially in the fields of criminology and urban sensing. Although Blondel et al. [1], Ghahramani et al. [2], and Calabrese et al. [18] made impressive contributions by examining mobile phone data applications in several studies, there is still a lack of research relating to mobile phone data in crime and land use. For example, Blondel et al. [1] reviewed social network applications that can be derived from mobile phone data in various disciplines and domains, such as urban sensing, epidemics, public transportation, and data protection. In addition, Ghahramani et al. [2] and Calabrese et al. [18] presented a survey of mobile phone data applications in the urban sensing domain. However, to the best of our knowledge, this is the first survey on the use of mobile phone data in crime and land use applications. No one had fully reviewed or discussed criminal applications for mobile phone data. This paper's key contributions are its discussions of various techniques and analytical perspectives that have been derived from mobile phone data to capture many aspects of human behavior, such as communication, social networking, and mobility patterns. The review also places a heavy focus on human attributes and characteristics that can be derived from mobile phone data by explaining the spatiotemporal and call features that have been extracted from mobile phone data to model human mobility and communication patterns and differentiate their functionalities. The review also sheds light on anomaly detection, churn prediction, and crime prediction applications, other topics that the literature lacks.

The remainder of the paper is organized as follows: in Section 2, we present the research methodology. In Section 3, we describe different types of mobile phone data. Mobile phone data applications in urban crime and urban sensing are discussed in Section 4. Section 5 discusses mobile phone social network applications. Section 6 discusses publicly available datasets, managerial implications, and methods. Section 7 provides potential research opportunities. Section 8 discusses privacy concerns, and Section 9 concludes the review.

2. Survey Methodology

This study reviews state-of-the-art methods and techniques regarding the use of mobile phone data in crime and urban research to generate the focus of the survey. In addition, the survey aimed to shed light on related studies or contributions made in mobile phone data research that the literature lacks, such as anomaly detection, churn prediction, and privacy concerns across various disciplines. Therefore, the survey was conducted using eight databases to search for relevant journal articles and conference papers. These sources include the following databases and digital libraries: ACM Digital Library, IEEE Xplore, MDPI, SAGE, Science Direct, Scopus, SpringerLink, and Web of Science. We ran keyword searches against titles, abstracts, or keywords in these data sources, allowing us to select the most relevant research journals and conference proceedings. The main keywords we used for our searches were "mobile phone data," "call detailed records," "call detail records," "call data records," "mobile phone datasets," "mobile phone networks," "mobile phone network data," "mobile network data," "mobile network activity," "mobile communication data," and "mobile phone call detail records." After that, we filtered the results for articles published between 2013 and 2021 according to our inclusion and exclusion criteria. This allowed us to choose 148 articles for review. The following are the inclusion criteria (IC) and exclusion criteria (EC) that were used to decide which articles to choose and which ones to leave out.

- IC1: A study has to be in a journal or proceedings
- IC2: Studies are peer-reviewed articles
- IC3: A study must be written in the English language
- IC4: A study must be published from 2013 to 2021
- EC1: Articles that are not written in English
- EC2: A study that is not published between 2013 and 2021

3. Mobile Phone Data Types

According to the literature [1,18,19], there are two types of mobile phone data: a type that records the details of an interaction between a mobile device and the network, known as event-driven mobile phone data, and another type, based on the cell tower location updates of mobile phones, known as network-driven mobile phone data.

In the first type, the data records the details of communication events that occur when mobile phones receive or initiate phone calls, text messages, or access the internet. The details of each communication event are held in a record that includes the IDs of the caller and the callee, the caller's connected cell tower ID, the callee's connected cell tower ID, the duration of the call, and a timestamp. Thus, this type is called "event-based mobile phone data," and it can be formed at individual and group (aggregated) levels. At the individual level, the data convey details about the communication and mobility behaviors of each party involved in the call; and at the group level (where the data are aggregated based on grouping a certain number of users in a certain area based on their connection to the cellular tower at various spatial and temporal scales), the aggregated data can provide relevant details including the user ID, the timestamp, and the cell tower ID for each of the individuals who are involved in the call.

The aggregated mobile phone data (usually called aggregated CDRs (call detail records), or mobile phone data at aggregated level) differs from the individual data (usually called mobile phone data at the individual level, or CDRs data) in the sense that they are anonymized and aggregated at different spatial and temporal scales. Thus, they are easier to collect and acquire, and their management is simpler compared to the individual data because they are aggregated based on the mobile phone devices that are connected to the BTS at different spatial and temporal scales. As a result, this allows for: the observation and monitoring of the spatial and temporal fluctuations of the population's activities at various scales (e.g., hourly, daily, seasonal) based on their movement patterns; the classification and clustering of users by parameters derived from location information, such as the most visited places or the most frequent locations that the users are present in; or to distinguish the residents of a place from visitors based on calling volume (the number of phone calls) that can depict the amount of activity in a given location. All of these measures can be acquired from the aggregated CDR data that offers spatiotemporal mobility patterns and communication behaviors. In contrast, the non-aggregated mobile phone data at the individual level (CDRs) requires permission from authorities for their collections and is difficult to manage due to the raw nature of the data, which requires cleaning and preprocessing steps before the data can be fully utilized. In addition, the raw data are hard to understand, and you need data mining tools to get useful information out of it.

Alternatively, the second type of mobile phone data aggregated at the cell tower level (which is also called "location-based mobile phone data", "passive mobile phone records", or "network-driven mobile phone data") primarily involves storing location information updates of mobile devices based on cell tower locations. In this type, the data records passive location updates of a mobile device from the base transceiver station (BTS) (also known as a cell, cell tower, or node B in 3G networks) at regular intervals and whenever the mobile device is switched on or off, receives a signal from the mobile network, or changes the type of connection. This type of data stores a record that details every event, including the user's ID, the timestamp, and the location (the cell ID), where each record is geolocated based on the nearest BTS the mobile device is connected to. Thus, in every geographical area, there is a defined number of BTS that cover the given area to ensure the quality of the communication services and optimize the signaling. Each of the mobile network events is recorded to a BTS through a base station controller (BSC) that controls and manages a set of BTS by governing the network traffic and performing network handover. Accordingly, whenever the signal between a mobile device and a connected BTS becomes weak, the BSC will hand over the connection to another BTS to guarantee an optimal transmission signal for that connection. This type differs from the first one in the sense that it neither depends on active events initiated by the user nor does it reveal communication (calling)

information about the receiving device and its location (i.e., the terminating user ID and the cell ID); thus, mobile phone data at cell tower level mostly captures spatial-temporal human mobility patterns due to the fact that only spatiotemporal information is recorded. Additionally, this type of data can record events either actively or passively based on how the mobile device interacts with the cell towers (passively, when the phone is turned on or off, receives a signal from the mobile network, changes the type of connection, etc.; actively, when a call is made, a text message is sent, etc.).

4. Human Mobility Patterns

Spatiotemporal information provided by mobile phone data can provide clearer insights into human movements and interactions in various applications. For instance, mobile phone data have been used to detect certain types of behaviors in cities and urban zones based on human mobility patterns, to investigate the relationship between human dynamics and crimes, to estimate population density, and to detect home and work locations. Therefore, analyzing human dynamics and movements is crucial to understanding their actions and activities.

4.1. Urban Environment

With the rapid growth and usage of mobile devices in urban environments, mobile phone data have been widely used as urban monitoring sensors, resulting in many diverse applications such as land use classification, estimating population density, home-to-work identification, urban hotspot detection, and others. So, looking into urban sensing applications is a good way to find out where and how much people do things in cities [2].

4.2. Classification of Urban Land Use

In mobile phone data studies, land use inference is based on human dynamics and activities as extracted from their mobility and call patterns. Spatiotemporal information extracted from human mobility patterns can, in this manner, be used to depict human activity patterns that shape land use types, with human activity patterns assumed to be linked to the time and location of recorded mobile phone usage based on the locations in which such activities take place [20]. Because of this, people's mobility and movement can stand in for many of the things they do, which can then be used to figure out how they use the land.

Empirically, human activity patterns based on mobile phone data are derived from specific characteristics provided by such data, specifically the temporal and spatial characteristics of mobile phone users as represented by concentrations of locations and times of mobile activities; for example, more activities occurring during the day (8 a.m. to 6 p.m.) on weekdays may imply that the main land use is for commercial or business areas, due to the fact that these types of areas have higher concentrations of mobile activities. Such human activity patterns as extracted from mobile phone data have been used to represent human activity patterns to infer land use over several studies, with most feature types being used, including spatiotemporal calling volume pattern features. This allows the capture of the hourly patterns of human activities across a given period of time. Human activity and mobility patterns are thus assumed to be reflected in call volume patterns, and such patterns can be captured when users make phone calls; their locations are recorded [21] and can thus be used to identify land use. Several studies have thus examined different features extracted from mobile phone data and used varying approaches to infer land use based on human activities.

Being able to capture human daily activities based on mobile phone data allows for basic classification of land use areas. Thus, several studies in this application have intended to apply classification and clustering methods to identify land use types. A semisupervised clustering fuzzy c-means (FCM) method was proposed to classify urban land use in Singapore into five land use types: residential, business, commercial, open space, and other [8]. The authors constructed a synthesized vector from a linear combination of two real vectors. These two vectors represent human activity patterns extracted from mobile phone data, showing the total calling volume and hourly call volume managed by each BTS. This was intended to overcome the gap left in previous studies where the time series of the call volume for two-day patterns were considered to infer land use types. The semi-supervised FCM was then used to classify the urban land use based on the synthetic vector. The same features were extracted from mobile phone data by [9] to represent human activities; however, they used a different approach whereby an unsupervised algorithm known as non-negative matrix factorization (NMF), which is well known for its use in dimensional reduction and feature extraction, was applied. The authors thus decomposed the call pattern matrix A, which contained only non-negative coefficients representing human activity patterns, into the product of two matrices, W and H (also composed of non-negative coefficients). After decomposing the call pattern matrix in this way, two basis vectors were identified, which illustrate the commercial/business/industrial (C/B/I) (work pattern) and residential (out-of-work pattern) characteristics of various areas. Mao et al. [13] used a similar approach to identify land use types with an extra feature extracted to depict human activities more precisely: temporal mobile phone call patterns were thus extracted to reveal different land uses along with total call volumes as aggregated at each cell tower. These two feature types helped to distinguish between those human activities that varied from weekdays to weekends. For example, the amount of time people spend at home during weekdays, based on features generated to represent the call volume by the hour of the day for each day of the week, showed that human activity in residential areas is up to three times higher at weekends due to the fact that people go to work on weekdays and stay at home more during weekends. Thus, during weekdays, activity reaches a peak in the middle of the day, between 12 p.m. and 6 p.m. in business areas. Yuan et al. [12] presented an unsupervised k-means clustering model that aimed to identify urban functional areas (UFAs) based on citizens' daily activities and communication activity intensity. To model this communication activity intensity (CAI), various spatial and temporal characteristics were extracted from mobile phone data, including the ratio of the number of calls made or received in a specific geographical area subdivision (GASs) during a specific time interval. Lastly, Furno et al. [22] used an unsupervised signature clustering algorithm to try to improve land use classification by combining mobile phone data with additional mobility information taken from GPS data, specifically the GPS traces of cars that were floating.

Using a supervised learning model to classify land use has been attempted by [11] and [23], with a sample of land use initially labeled to train the classifier. Zinman and Lerner [23] applied a random forest (RF) algorithm to classify urban areas in Tel Aviv by extracting two types of features previously used across the literature [8–10,12,13] along with additional feature types representing communication habits to capture human communication behaviors, such as call duration, contact type (phone calls, accessing the internet), the weekly pattern features based on differences in communication activity between weekdays and weekends, and contact features such as the average number of days on which people engaged in or performed cellular contact in a given cell over one hour. Similarly, [11] applied a support vector machine (SVM) classifier algorithm, one of the most popular supervised learning algorithms, to classify urban land use types in Beijing. These were split into six classes: (a) residential, (b) business, (c) scenic, (d) open, (e) other, and (f) entertainment. As SVM is a supervised ML approach, it requires training datasets with land use labels to train the model. Thus, based on the fact that mobile phone data are not inherently labeled data, the authors selected land use labeled samples from Google EarthTM at a resolution of 2776 pixels as the training samples to train the appropriate classifiers. During the process of putting together human activity patterns from mobile phone data, the authors used the same spatiotemporal characteristics to show human activity patterns as those most often used in previous studies [8–10,12,13].

A different approach based on community detection techniques (community graph clustering), which is different from traditional clustering techniques such as K-means, FCM, and hierarchical cluster analysis (HCA) to detect land use types, was adopted by [14].

Lenormand et al. [14] wanted to make activity profiles based on the weekly activity that show the number of mobile phone users per hour during the days of the week to find four land use types (residential, business, logistics/industrial, and nightlife) in Spain.

As a way to improve land use classification outside of classical clustering techniques, Ríos and Muñoz [15] proposed a new clustering technique that aims to detect land use in Santiago City. This was based on adopting a latent variable clustering technique.

Other studies have attempted not only to classify land use but also to investigate additional information about the relationships between human activities and urban land use and how different urban land use types can influence or impact people's lives. Jacobs-Crisioni et al. [20] aimed to investigate the impact of mixed and dense land use on urban activity dynamics in Amsterdam, Netherlands, to determine whether land use density and mix can prolong high levels of activity in urban areas; they thus extracted spatiotemporal characteristics from mobile phone data to depict human activity levels and then applied spatial regression models. The results showed that urban areas with high attraction levels corresponded with increased urban activity intensity, supporting the hypothesis that mixed land use diversifies urban activity dynamics and increases profiling activity intensity. Put simply, mixing shops, businesses, and meeting places has an additive effect on activity levels. Yang et al. [24] aimed to explore the relationship between human convergence and divergence patterns extracted from mobile phone data and land use characteristics (commercial, industrial, residential, public, and transport land). To achieve such a goal, the authors first extracted spatial and temporal characteristics provided by mobile phone data to depict human convergence-divergence patterns and then applied multinomial logistic regression (MLR) to reveal the effects of land use characteristics on human convergencedivergence dynamics. Liu et al. [25] used commuting flow patterns (origin-destination (OD) commuting flows) to depict human activities with the aim of exploring urban land use types as well as investigating whether urban land use influences commuting flows. By using a Louvain modularity-based algorithm, Novovic et al. [26] attempted to apply community detection techniques based on users' dynamics and activity variations over space and time to infer the correlations between human dynamics and land use. The idea of modularity has been used a lot in research on mobile phone data as a good way to measure how good a partition is.

The variation in researchers' findings on land use can be attributed to different perspectives on how to analyze mobile phone data in order to depict human activities. For example, individuals' behaviors have been interpreted in diverse ways, including, among others, user dynamics, human activities and mobility patterns, commute flow patterns, temporal activity patterns, and human convergence/divergence dynamics. These multiple analytical perspectives have been developed to deal with the diverse spatiotemporal and call features extracted from mobile phone data, such as spatiotemporal call volume, users' daily trajectories, communication habits, weekly patterns, and contact features.

Table 1 shows the different analytical perspectives and features extracted, as well as the methods used to infer land use. Previous studies have shown that detecting land use is best handled as a clustering problem to which algorithms can be applied, while other statistical methods have been used to uncover the relationships between human activities and land use.

Reference	Analytical Perspective	Feature/Characteristic	Application	Description	Algorithm/Technique
[8,9,11–13]	Human activities and mobility patterns	Spatiotemporal call volume: total and hourly call volume managed by each base transceiver station (BTS) (i.e., total number of calls or mobile phone devices managed by a given BTS over a given period)	Classification of urban land use types	These studies have depicted human activity patterns based on extracting spatiotemporal call volume features	FCM [8], NMF [9,13], SVM [11], and k-means [12]
[13–15]	Temporal changes in human activities	Temporal call patterns and volume: calculations or estimations of the number of calls or mobile phone devices managed by each BTS tower every hour in a seven-day week (i.e., weekdays and weekends)	Land use detection	These studies have detected land use patterns based on temporal changes in human activities to capture human behaviors' variation over time (e.g., human activity trough in the middle of the day on weekends)	NMF [13], community detection algorithms [14], and latent Dirichlet allocation [15]
[26]	Human dynamics	Spatiotemporal features: cell tower identification that shows BTSs' exact location and aggregated mobile network traffic activity for each BTS at 10-min time intervals	Investigation of relationship between human dynamics and land use	This study investigated the correlations between land use and human dynamics, depicting human dynamics as a graph in which nodes are BTS towers and edges represent communication traffic between two nodes	Community detection algorithms
[25]	Human commuting patterns	Users' daily trajectories based on spatiotemporal features: users' location represented by cell tower location (e.g., a residence location can be identified based on the most frequently used cell tower locations during the night hours)	Clarification of relationship between commuting flows and variables such as industrial, commercial, residential, and educational land use	This study's main goal was to gain a fuller understanding of the relationship between land use variables and commuting flows, so a gravity model was used (i.e., a widely used technique for assessing commuting flow patterns), which shows that commuting between two locations i and j, with origin population mi and destination population mj, is proportional to the product of these populations and inversely proportional to a power law of the distance between them [25]	Gravity and regression models
[23]	Human daily and weekly activity patterns	Spatiotemporal call features: spatiotemporal call volume (i.e., total call volume and compared call patterns), communication habits, weekly patterns, and contact features	Land use classification	This study focused on various features to capture many aspects of human activity patterns and depict variation in human activities on weekdays and weekends	RF

 Table 1. Prior research on human activities and land use based on mobile phone data.

Investigating densely populated urban environments where criminal activities are much more likely to occur has long been a popular topic in mobile phone data research. The structure of built urban environments can be seen in many attractions and centers of activity such as bars, clubs, hotels, shops, and schools that draw many people and create opportunities to commit crimes as these places are known for being crowded. Urbanization pulls multiple individuals to cities and thus results in more crime [27]. Empirical research has shown that urban environments where people concentrate and participate in daily activities, for example, schools, stations, shopping centers, sports venues, and entertainment areas, can be defined as crime generators. Crime attractors, in turn, are places that are not crowded but attract people who want to commit crimes [28], such as bars, nightclubs, automatic teller machines, and banks [27].

Mobile phone data have been used to prevent, fight, and deter urban crime by predicting crime hotspots, detecting and identifying suspects and criminals, and investigating the relationships between human mobility patterns and criminal activity patterns. The digital tracks left by people at locations where a crime has taken place can reveal a representative sample of the population present at the crime scene at a given time, thereby providing insights into correlations between individuals' mobility and criminal activities. Mobile phones connect to a given cell tower located in the area where a crime has taken place, leaving digital traces or evidence that can be used to conduct empirical research on crime patterns and the dynamics of criminal behavior from a people- and place-centric perspective.

The literature reports various crime-related applications that have been derived from mobile phone data aggregated at the cell tower level. For example, some studies [29,30] have identified crime hotspots in London by extracting human mobility patterns from mobile phone records. To depict these mobility behaviors, the cited authors focused on spatiotemporal features, such as the total numbers of calls or mobile phone devices for each cell tower every hour. Other researchers [27,28,31–33] have similarly relied on defining specific populations' spatiotemporal patterns to investigate the relationships between human dynamics and spatiotemporal crime patterns. These studies have tracked features of ambient populations that indicate specific areas are at risk of criminal activities or that individuals are in danger of becoming victims of theft or assault, thereby confirming that ambient populations' configurations have a significant impact on crime patterns and rates.

Scholars have discovered additional applications by analyzing mobile phone data at the individual level, or CDRs. For example, previous studies [34–38] have detected or constructed criminal social networks based on criminals' communication behaviors. To depict this type of communication, the cited authors have extracted call features such as the duration of calls between two criminals, the frequency with which they make calls, the maximum, average, and minimum duration of these calls, and the maximum number of outgoing phone calls or messages. These features then allow these researchers [34–38] to represent criminal organization networks that show whether nodes (i.e., criminals) are highly influential or less influential members. The results can, in turn, help criminal investigators understand criminal networks' hierarchical structure, as criminal investigators have trouble determining who belongs to criminal organizations, who heads them, and what relationships exist within them due to the nature of the raw data available.

Still, other studies [4,39,40] have built forensic detection models to separate suspects from non-suspects. The cited authors have analyzed criminal communication behaviors by extracting suspects' call features such as timestamp, frequency, and average, maximum, average, and minimum duration. Table 2 presents crime-related applications based on mobile phone data, the various features extracted to depict human behaviors, and the methods used.

Reference	Analysis Perspective	Feature/Characteristic	Application	Description	Algorithm/Technique	Geographical Unit/Spatial Unit
[29,30]	Human mobility patterns	Spatiotemporal features: cell tower IDs and timestamps to calculate the total number of mobile phone devices in each cell tower every hour	Crime prediction	Human mobility patterns extracted from mobile phone data can be used to predict crime hotspots	RF	Cellular network cells: 124,119 cells
[41]	Human mobility patterns	Spatiotemporal features such as cell tower IDs and timestamps to estimate footfall count entries in each cell per hour	Crime prediction	The results show that the relationship between crime activities and the diversity of the ages and ratios of visitors negatively correlated	Correlational analysis: Tjostheim's coefficient	Grid cells: the geographic area is divided into 23,164 grid cells.
[42]	Human daily mobility patterns or daily population mobility patterns	Extracted spatiotemporal features: cell tower IDs and timestamps	Crime prediction	The daily mobility flows of the general population have been captured to provide a template of the daily mobility of criminals	Regression analysis: conditional logit discrete choice models	Census units: 1616 census units
[33]	Human mobility patterns and social activities	Spatiotemporal features and call logs: cell tower IDs, timestamps, and the number of phone calls or short message service (SMS) made and received	Crime prediction	Mobile phone data have been used to measure the ambient population at risk, and results showed a strong correlation between ambient population and criminal activities	Correlation analysis: Moran's I statistic, and regression: negative binomial regression analysis	Grid cells: the study region is partitioned into equally sized grid cells of (306×306 m).
[28]	Mobile phone activity	Spatiotemporal features: timestamps and cell tower IDs to estimate or count the number of times a mobile phone device communicates with the cell tower, which this parameter has later used to measure the size of the ambient population	Crime prediction	The results showed strong correlations between the ambient population measures (workday population, mobile phone data, and population 24/7 daytime estimates) and crime patterns (the crime of theft from person)	Correlation analysis: Spearman's rank correlation coefficient [ρ] statistics	Lower super output areas (LSOAs): cellular network grid cells converted to LSOA geographical units

Table 2. List of applications, methods, and features used in criminology based on mobile phone data.

|--|

Reference	Analysis Perspective	Feature/Characteristic	Application	Description	Algorithm/Technique	Geographical Unit/Spatial Unit
[31,43]	Mobile phone activity	Spatiotemporal features: timestamps and cell tower IDs to calculate the total number of mobile phone devices in each cell tower every hour over a 3-month period	Crime prediction	A stronger correlation was found between ambient population and crime rates	Correlation analysis: Pearson correlation coefficient and point-biserial correlation coefficient	Grid cells of 200 \times 200 m
[27]	Human mobility patterns	Spatiotemporal features: timestamps and cell tower IDs	Crime prediction	The results demonstrate a negative relationship between ambient population and street robbers' criminal activities, in which ambient population has a significant effect by reducing opportunities to commit crimes	Correlation and regression analysis: discrete choice models and negative binomial regression	The geographical areas were created using Thiessen polygons, where 52,026 cell towers were mapped onto polygons
[44,45]	Intra-daily mobility patterns of the population	Spatiotemporal features: timestamps and cell tower IDs to identify the origin and destination of each user	Crime prediction	These studies proposed a new measure in calculating crime rates and exploring crime patterning, which is the exposed population at risk, which includes a mixed population of, for example, criminals, victims, and guardians. The results showed that the exposed population is more significant than the ambient population in exploring violent crimes in public spaces	Correlation analysis: Spearman's rank correlation coefficient (ρ) statistics [44]. Regression analysis: negative binomial regression model (NBM) [45]	Lower super output areas: 1673 LSOAs

Table 2. Cont.

Reference	Analysis Perspective	Feature/Characteristic	Application	Description	Algorithm/Technique	Geographical Unit/Spatial Unit
[32]	Daily movement patterns of migrant and native offenders	Spatiotemporal features extracted: timestamp and cell tower ID to count the number of mobile phone devices connected to a given cell tower on a per-hour basis. This feature helps to estimate ambient population and criminal movements when a crime takes place	Detecting criminal mobility patterns	The results show that the ambient population has a positive relationship with dynamic patterns of violent crimes committed by migrant offenders	Descriptive statistics and negative binomial regression models	The geographical areas were shaped using the Thiessen polygon technique, where 52,026 cell towers were represented as Voronoi cells
[5]	Spatiotemporal mobility patterns of terrorists	 Spatiotemporal features of terrorists: (1) variation over time: this feature depicts the mobile phone spatiotemporal patterns of criminals over a given period and then allows to distinguish criminal mobility patterns and activities that contain varied locations over time (2) frequency: number of times a given cell tower has visited or contacted by criminals (3) distance: the distance between the locations where the criminals visited and their homes 	Detecting mobility patterns of terrorists	This study identified the meaningful places for criminals based on the digital traces they left at home and other visited locations. The traces were then analyzed to determine the changes in the terrorist's spatial behaviors	Correlation analysis: Spearman's rank coefficient (ρ) statistics, Pearson's correlations, and statistical analysis: the cumulative distribution function	Cellular network cells: cell tower locations were spatially approximated to the postcode area, which in the United Kingdom covers a small area of approximately 0.14 km ² .

Table 2. Cont.

Reference	Analysis Perspective	Feature/Characteristic	Application	Description	Algorithm/Technique	Geographical Unit/Spatial Unit
[3,34– 37,46,47]	Criminal communication behaviors	Call features: outcoming/incoming calls, call frequency, maximum and minimum numbers of incoming or outgoing calls and messages, call timestamps, temporal changes in mobile phone call patterns, caller ID, called ID, type of communication (phone call, SMS, MMS, or voice), and call duration	Detecting criminal networks	These studies built multiple forensic systems to detect criminal networks based on their calling characteristics. Here, a criminal network is represented by a set of nodes (criminals) and the edges or links between them represent a communication (i.e., a phone call or SMS)	Social network analysis tools and graph algorithms such as Prim's minimum spanning tree algorithm [35], the Girvan-Newman algorithm [34], Space algorithm [3], Blondel's community detection algorithm [4], and Fruchterman-Reingold algorithm [47]	N/A Missing location data (i.e., the geographical position of nodes is unknown)
[48–52]	Communication and mobility patterns of suspects	Spatiotemporal and calling features: the SIM numbers and location ID of the suspects, calls made between the suspects, maximum call duration, call frequency, phone calls made at the crime location, the most frequent caller, the number of times the suspect called other suspects, suspect trajectories, and others	Identifying suspects and their associates	These studies built a call detail record query system to detect suspects and suspicious groups.	Big data technologies and analytics such as Hive, Hadoop MapReduce, and the Hadoop Distributed File System	The coverage areas of cell towers have not been intersected with any geographical units.
[4,39]	Suspects' communication behaviors	Call features: call duration between suspects, maximum and average call duration, maximum duration of outgoing and incoming calls, standard deviation duration of incoming calls, phone calls made at the crime location, and others	Suspect classification	These studies built suspect classification models based on machine learning approaches that can classify suspects from non-suspects	Bayesian network [39] and graph convolutional networks [4]	N/A

As noted in Table 2, most researchers have used statistical methods such as correlation and regression analysis in crime prediction, and SNA tools have been widely used to detect criminal networks. In suspect identification, big data technologies are the most frequently used. Moreover, the variation in research on crimes can also be viewed from various analytical perspectives, and features extracted from mobile phone data can be used to depict human activities, as illustrated in Table 2 and Figure 1.



Figure 1. General concept of criminal activity detection.

In addition, some studies applied spatial mapping techniques to intersect or project mobile phone locations from mobile network cells [53] into spatial units, such as census units [42] and Lower Layer Super Output Area (LSOAs) units [44,45].

However, other studies [48–52] have not considered the spatial mapping of mobile devices to a given spatial unit. In this case, mobile phone locations were originally assigned to the coverage areas of a base station [16]. Consequently, undoing the spatial mapping of mobile phone presence will affect the accuracy of the sampling population at or near crime scenes and can lead to inaccurate detection results.

4.4. Public Health

Mobile phone data have been used in public health research to fight against infectious diseases, understand human mobility after natural disasters, measure and estimate human mobility in relation to the epidemiology of infectious diseases, and quantify exposure to air pollution [54]. These above-mentioned multiple health applications have been derived from human mobility patterns by extracting spatiotemporal characteristics of individuals from mobile phone data. Indeed, many aspects of human daily activities and lifestyles are linked to human mobility patterns, and measuring and monitoring human mobility patterns to public safety and human health. For example, [55] depicted individual mobility patterns by extracting their spatiotemporal features to examine and quantify their exposure to air pollution.

Recently, human mobility patterns and interactions have allowed a better understanding of COVID-19 trends and geographic distribution [56]. This is one of the important roles that mobile phone data has played recently in controlling and preventing COVID-19 and tracking population movements. During the coronavirus disease 2019 (COVID-19) pandemic, many COVID-19 applications were built using data from mobile phones. These applications were used, for example, to help public health response to COVID-19 [57], to track changes in people's mobility patterns [58], to control how the COVID-19 pandemic spreads [59], and to find out if there is a link between changes in mobility patterns and the number of daily COVID-19 cases [60].

So, being able to measure and track people's movements can help reduce health risks [61] and people's exposure to air pollution, which can cause health problems such as breathing and heart problems [62]. Table 3 presents public health applications based on mobile phone data, analysis perspectives, and study findings.

Reference	Application	Feature	Study Finding	Analysis Perspective
[62]	Dynamic estimation of individual exposure to air pollution.	Spatiotemporal	The results show that exposure to nitrogen dioxide (NO ₂) goes up by 4.3% during the week and by 0.4% on the weekends. Due to the fact that, during the week, people who live in small towns near big cities are exposed to more NO ₂ because they work in these cities	People's travel patterns
[63]	Estimation of human exposure to air pollution	Spatiotemporal	The results indicate that the home-based method (HBM), which assumes that all individuals spend the entire day at their homes (individuals who are not highly mobile), is still a useful measure for estimating their exposure to air pollution	Daily mobility patterns
[64]	Controlling the spread of dengue fever	Spatiotemporal, SMS	The results show that the spatially targeting SMS policy that encourages people to avoid and cancel trips to high importation risk areas can help to reduce the risk of disease spread	Spatiotemporal travel patterns
[65–67]	Controlling and measuring the spatial spread of malaria	Spatiotemporal	These studies prove that mobile phone data are effective in controlling and estimating the spread of malaria parasites by analyzing the mobility patterns of individuals in countries such as Bangladesh, South Africa, and Madagascar	Human travel patterns
[60]	Investigation of the correlation between mobility patterns and COVID-19 cases	Spatiotemporal	The results indicate that a decrease in human movement (a reduction in the number of individual trips) is associated with a decrease in the growth rate of COVID-19 cases	Human daily mobility patterns
[68]	Real-time predictions of human movement during the Tokyo earthquake	Spatiotemporal	The proposed assimilation method yields encouraging results for estimating the real-time movement of people during earthquakes	Real-time human movement

Table 3. List of applications used in public health.

As noted in Table 3, several applications have been examined in health research. It can be observed that travel and mobility patterns have been widely used to depict human activities, allowing investigation into the effect of travel and the transmission of infectious disease and estimating the number of trips to areas with a higher risk of COVID-19.

4.5. Transportation Research

Due to the fast growth of telecommunication networks, a large quantity of data about how people move and behave in space and time is being generated, which could be used to evaluate and analyze the travel patterns and social interactions of the total population [69]. Understanding the movement of people and where they live is essential because it can help urban planners and organizers manage traffic flow and plan public transportation services [2]. In this matter, mobile phone data have played an essential role in depicting people's mobility and social interactions in urban areas due to the wide usage of smartphones. The digital traces (e.g., mobile phone data) left by this large number of devices provide valuable information that facilitates the understanding of passenger travel behaviors, human movement behaviors in urban areas, and changes in human driving behaviors during road traffic. Because of this, knowing how people travel and act could lead to ways to improve the quality of life in a city on a large scale.

However, one of the challenges in transportation research is to understand the travel behavior of passengers with regard to identifying the hub passengers and detecting their transport modes (e.g., train and subway). Analyzing the transportation hub, which is the transition point for passengers to switch between various types of transportation (train and subway), is necessary in order to understand passenger travel demand [70]. This ultimately helps to support and evaluate urban transportation planning and management [71].

The second challenge on which transportation literature widely focuses is enhancing the detection of the transport mode of passengers. This difficulty is due to the nature of mobile phone data, which is noisy, sparse, and irregular. Several techniques have been proposed to improve the performance of transport mode detection. For example, Graells-Garrido et al. [72] aim to identify the distribution of usage of transportation modes in Santiago, such as the metro, bus, and car, by building a pipeline method for inferring trips based on user trajectories. Chin et al. [73] aim to improve the detection of transport modes by applying multiple supervised and unsupervised detection algorithms such as rule-based heuristics, RF, fuzzy logic, and partitioning around medoids (PAM). The author in [74] aim to identify travelers' transport modes, such as buses, cars, and railways, based on detecting travel speeds for each transport mode. For example, if a trip's travel speed exceeds 15 km/h and there are no subway or bus stops within 500 meters of its source or destination, the trip is designated as a car trip [69]. Finally, Bachir et al. [19] aim to estimate total origin-destination flows by modes of transportation, such as road and rail, by combining five different types of mobility datasets that allow for fine-grained daily population dynamics.

There are also other problems, such as the fact that existing mode detection techniques are mostly made to recognize easy-to-detect modes (subway, train) or more general mode groups (e.g., rail versus road, moving versus stationary) [69,73].

The literature shows that mobile phone data have been used for many transportation research purposes, such as estimating passenger origin–destination flows, predicting traffic congestion, and other applications. Table 4 presents transportation applications.

Reference	Application	Description
[75]	Traffic congestion detection	The study aims to classify traffic conditions into high, medium, and low traffic levels based on handover records that show the number of handover events in a cellular tower
[76]	Origin-destination flow estimation	The study aims to study traffic flow in Dhaka city by constructing origin-destination (OD) matrices based on phone users' trajectories
[77–79]	Urban planning and management	These studies aim to investigate and understand the dynamics of human mobility and human travel patterns in urban areas, which paves the way for improving traffic planning, public transport design, and transportation infrastructure design
[19,80,81]	Transport mode detection	These studies analyze the travel behaviors of passengers in order to identify the modes of transportation that passengers take, such as metro, train, car, and bus

Table 4. List of transportation applications in mobile phone data.

5. Human Communication Behaviors

Mobile phone data at the individual and aggregated levels, usually called CDRs and aggregated CDRs, respectively, can be used to investigate and study human communication behaviors and social communication patterns due to the fact that mobile phone data at these levels contain communication information. Mobile phone data aggregated at the cell tower level does not contain details of the other side of communication (such as the callee ID or the callee's connected cell tower ID) and thus cannot be used in applications regarding human communication behaviors. So, CDRs have been used to study a wide range of human social-related topics in many different ways. For example, it has been used to infer social ties and relationships, find social networking communities, find criminal relationships, identify suspects and criminals based on their calling patterns, and do other things that have to do with how people talk to each other.

5.1. The Construction of Social Networks

Mobile phone network data can be used to reconstruct social network interactions [1], which depict social structures as mobile phone customer interconnections. Most of the time, this is achieved through the use of mobile phone data as a graph consisting of a set of nodes and a set of edges.

Generally, any social network can be represented by a graph, *G*, on a set of nodes, *V*, and a set of edges, *E*. To construct a social network of criminals based on communication behaviors extracted from CDRs, calling characteristics must be extracted, such as the number of calls or SMS messages, duration of calls, and timestamps. Empirically, mobile phone data can create a social network based on individuals (subscribers) making or receiving calls or messages; these individuals are classified as actors (nodes) within the network, and the links between the actors represent the various types of communications (e.g., calls or messages). In a real-life scenario detecting a criminal social network, Taha and Yoo [35,36] created a criminal social network derived from mobile phone data that showed a criminal network consisting of 62 nodes representing all individuals involved in the incident; this contained 153 edges, where these edges represented various reported interactions (such as calls and SMS messages) between the actors (criminals) involved in the incident. However, in studies of mobile phone data, different ways to build or make social networks have been found. These ways vary from one study to the next based on the purpose of the study and the features taken from mobile phone data.

In addition, different network metrics (see Section 4.2 for more details) may be applied during any network analysis, such as the use of a centrality measure or metric to find the most influential subscribers in the network or the use of a reciprocity measure [82] to analyze linking behaviors in a network by determining whether they demonstrate transitive behaviors between nodes [83]. Yet there are also other network metrics that are used for different things, such as figuring out how a network is put together or figuring out how strong the modules in a network are, such as network modularity and efficiency. Table 5 shows some studies that analyze mobile phone data based on social network analysis (SNA) tools.

Reference	Application	Feature	Algorithm	Network Measure
[34,47]	To detect criminal networks	Call features	Girvan–Newman and Fruchterman–Reingold	Degree centrality, eigenvector centrality, closeness centrality, transitivity, betweenness centrality, and transitivity
[3,35]	To detect criminal networks	Call features	Concept space approach and Prim's algorithm	Vertex-centric, edge-centric
[84]	To detect customers who are likely to fail to pay their mobile bill	Call feature and spatiotemporal features	SLPA	Closeness centrality and reciprocity measures
[85]	To detect users' social interactions	Call feature	Bron and Kerbosch's	Persistence, disparity, and reciprocity measures
[86]	To detect ethnic communities in Ivory Coast	Call and spatiotemporal features	Louvain	Asymmetries and assortativity coefficient
[87]	To detect human spatial interactions in China	Spatiotemporal features	Infomap, Louvain, and REDCAP	Degree, strength, rich-club coefficient, and assortativity coefficient
[88]	To detect socio-economic groups in Ivory Coast	Call and spatiotemporal features	Louvain	Rich-club coefficient and PageRank
[26]	To detect he spatial interactions of communities in Milan, Italy	Spatiotemporal features	Louvain	Betweenness centrality, degree centrality, and PageRank
[89]	To detect individual 's spending behaviors	Spatiotemporal features	Louvain	Diversity, radius of gyration, and homophily
[90]	To detect socio-economic communities in Santiago, Chile	Spatiotemporal features	Louvain	Segregation measures (i.e., isolation metric)
[91]	To detect urban communities in Dublin, Ireland	Spatiotemporal features	Louvain	Newman's modularity metric

Table 5. Summarizes social network applications, algorithms, and network metrics.

5.2. Network Metrics

Degree centrality: The number of direct links that a node possesses can be defined as its degree of centrality. Any nodes with a high degree of centrality may be regarded as hubs (crucial channels of communication). Equation (1) can be used to calculate degree centrality, in which *n* represents the number of vertices in the network, whilst x_{vu} will be equal to 1 if vertices *v* and *u* are linked. If not, this figure will be 0 [3].

$$C_I(u) = \sum_{v \neq u}^n x_{vu} \tag{1}$$

Betweenness centrality: This metric assesses the distance between a specific node and other nodes in the same network. These intermediate elements can exert strategic control and influence on other elements [34]. This centrality measure is largely based on the assumption that any node lying between the shortest path connecting two nodes is central. People who have a high degree of centrality are thought to act as "gamekeepers" in the network. Shortest paths: The number of shortest paths between vertices passing through the vertex can be used to calculate its betweenness centrality [3]. Equation (2) can be used to work out the betweenness centrality, in which σ_{st} represents the number of shortest paths between vertex *s* and vertex *t*, whilst $\sigma_{st}(u)$ represents the number of shortest paths between vertex *s* and vertex *t* passing through *u*.

$$C_B(u) = \sum_{s,t \in G, s \neq t} \frac{\sigma_{st}(u)}{\sigma_{st}}$$
(2)

Closeness Centrality: Closeness refers to the length of the shortest path compared to all other vertices. In other words, it assesses the proximity of a vertex to other vertices. Equation (3) can be used to measure closeness centrality, in which (ui-, uj) represents the distance between vertex ui and vertex uj.

$$c_{c}(u_{i}) = \frac{1}{\sum_{i \neq i}^{n} d(u_{i}, u_{j})}$$
(3)

Identifying the influential members in criminal networks by assigning weight to a vertex: Here is an initial formula proposed by [36] that can be used to assign each vertex v_k with a weight that accurately represents its importance in the network compared to other vertices. To calculate the weight of k, the following factors must be considered: (a) The frequency of incoming and outgoing communications to v_k (i.e., the number of times vertex k appears in reports of criminal activity compared to other vertices). (b) The frequency of incoming edges to v_k (i.e., the number of vertices with edges outgoing to k, as well as the number of vertices with edges incoming from v_k). The following equation was employed by the researchers to calculate the relative weights.

$$w(v_k) = \frac{\sum_{i=1}^{|v_k^E(in)|} |(v_i, v_k)| + \sum_{j=1}^{|v_k^E(out)|} |(v_k, v_j)|}{\sum_{i=1}^{|v_k^E(in)|} 0.8|v_i, v_k| + \sum_{j=1}^{|v_k^E(out)|} 0.6|v_k, v_j|}$$
(4)

6. Discussion

The study results show that various spatiotemporal and call features and characteristics have been extracted from mobile phone data to depict human activity patterns. These characteristics vary between studies based on their purposes and the aspects of human behaviors that they aim to investigate (i.e., mobility patterns, social interaction, communication, etc.). For example, to identify suspects, [48,50] extracted call features, such as phone calls that suspects made or received at crime scenes, using them as evidence of their involvement in the crimes. Similarly, [39] identified suspects based on call features but with additional features such as "average call duration with suspects," "maximum call duration," "average call duration," "minimum duration of a call out," "maximum duration of an incoming call," and "standard deviation duration of an incoming call." In other applications, [8,9] explored human activity patterns to infer urban land use based on spatiotemporal calling volume patterns. Similarly, [23] inferred urban land use based on spatiotemporal calling volume patterns and additional features, such as communication habits, weekly patterns, and contact features. Here, we notice that multiple features have been extracted to depict human behaviors, relying on either spatiotemporal or call features. However, there is currently no standardized way in which specific features should be extracted, as the literature demonstrates multiple approaches. There is also no scientific evidence on which spatiotemporal and call features are better than others when it comes to accurately and effectively reflecting on and depicting human behaviors.

In this section, we also discuss the availability of mobile phone datasets for public use, the distribution of mobile phone data applications, machine learning methods, and managerial implications.

6.1. Public Datasets in Mobile Phone Data

As Table 6 states, in the literature, many studies have made their datasets privately available, especially mobile phone data at the individual level (CDRs), due to privacy concerns, while open-source public datasets are limited and mostly suitable for specific applications related to human mobility patterns due to the absence of communication information.

Dataset	Description	Application	Limitation/Link
Nodobo mobile phone records dataset	Nodobo contains mobile phone records of 27 graduating high school students from September 2010 until February 2011. These data were collected by a group of researchers at the University of Strathclyde, United Kingdom	Used in applications regarding detecting criminal networks [3,48] and anomaly detection [92]	It contains only communication data; hence, it is suited for applications related to human communication behaviors (https://pureportal.strath.ac.uk/ en/datasets/nodobo-mobile- phone-usage-data (accessed on 20 September 2022))
Smartsteps dataset	Anonymized and aggregated human behavioral data derived from Telefonica Digital Company, Portugal.	Used in applications regarding crime prediction in urban areas [29,30,41]	It contains only spatiotemporal data (https: //tis.smartsteps.telefonica.com (accessed on 10 August 2022))
Telecom Italia dataset	This mobile phone data published by Telecom Italia in 2014	Used in applications regarding land use detection [22], urban hotspot detection [93], anomaly detection [94], and mapping population density [95]	It Is only suitable for limited applications (http://www.telecomitalia.com/ tit/en/bigdatachallenge/contest. html (accessed on 12 November 2022))
Orange's 'Data for Development' (D4D)	This dataset contains two types of mobile data, mobile phone data at the aggregated level (aggregated CDRs) and the cell tower level. These data were generated from Orange mobile phone operator in Ivory Coast	Used in applications regarding home/work detection [96], land use detection [9], identifying user habits [97], and unusual event detection [21]	Only suitable for limited applications (http://www.d4d.orange.com (accessed on 21 October 2022))
OpencellId database	This is the largest open database of cell towers in the world	Estimating ambient population [98]	It contains only location data (https://opencellid.org/ (accessed on 14 November 2022))

Table 6. List of publicly available datasets used in the literature.

6.2. Distribution of Mobile Phone Data Applications

Research on mobile phone data has been widely used in various applications in different domains because even the smallest bit of information is enough to trigger many new applications [1]. Although some studies on mobile phone data are used to develop their unique applications [99], most studies share a relatively broad classification. Figure 2 illustrates different applications of mobile phone data.



Figure 2. Network visualization of applications in mobile phone data.

The data used in this study were collected from the Scopus database, and VOSviewer software was used to obtain a visual representation of the data. In the network presented in Figure 2, there are three clusters. The first cluster, the red cluster, contains nodes that represent applications derived from human activities and mobility in urban sensing, such as urban transportation, travel behaviors, urban dynamics, estimating population densities, and estimating origin–destination matrices. The nodes in the second cluster, the blue cluster, represent human communication behaviors, and social activities, such as social networking, mobile networking, 5G mobile communication, crime networks, churn prediction, and anomaly detection. The third cluster, the green cluster, contains nodes related to epidemiology and public health, such as health risk assessment, estimating physical distancing during pandemics, COVID-19, disease control and prevention of disease transmission, estimating air pollution, and estimating individual exposure to air pollution. We noticed some new and emerging applications in which mobile phone data have been used to monitor physical distancing and model the spread of COVID-19 for disease control and prevention.

6.3. Methods

Table 7 shows that various machine learning models and algorithms have employed mobile phone data to classify, cluster, and predict human communication behaviors and mobility patterns in various contexts. Most studies have used machine learning approaches to solve classification and clustering problems. Machine learning (ML) is a subset of artificial intelligence that can learn from data and make decisions with little or minimal human intervention [100]. On the other hand, deep learning (DL) is a subset of machine learning that can automatically learn useful features and representations from raw data and output results without human intervention [101]. However, DL algorithms require significantly more data than ML algorithms to function properly. Due to its complex multilayer structure, a DL system requires a large dataset to eliminate fluctuations and produce high-quality interpretations.

Table 7. Machine learning (ML) and deep learning (DL) models and algorithms used by previous researchers in mobile phone data studies. The abbreviation of algorithms is shown in the back matter.

References	Algorithm/Model	Objective
[102]	SVM, NB	To classify user relationships
[8]	FCM	To classify urban land use in Singapore
[4]	GCN	To classify criminals from non-criminals
[39]	BN	To classify suspect users from non-suspect users
[103]	GBDT	To detect significant locations in users' visiting patterns
[29,30]	RF	To classify geographical areas into two classes, high or low crime levels
[7]	RF	To predict population density in Portugal and France
[23]	RF	To classify urban areas in Tel Aviv
[104]	DBSCAN, GMM	The DBSCAN algorithm is used to cluster users' trajectories into meaningful places, while GMM is used to identify users' habits
[11]	SVM	To classify urban land use in Beijing into six classes, (a) residential, (b) business, (c) scenic, (d) open, (e) other, and (f) entertainment
[12]	K-means	To identify urban functional areas (UFAs) in Beijing
[105]	GAN	To create artificial maps of population density distributions
[106,107]	K-means	To classify city users based on their calling behaviors into different types of city geographic areas, including residents, visitors, and commuters
[108]	MLP	To predict the real estate price in Budapest, Hungary
[109]	RF, GBDT, SVM, Adaptive boosting	To reconstruct individual trajectories
[110]	MLP, CNN, LSTM	To predict crowd distributions of people in urban areas
[111]	NB, LR, RF, DT, KNN	To prompt or recommend the best mobile phone contract services based on customer communication behaviors
[112]	BP	To estimate individual exposure to particulate matter (PM2.5) air pollution
[113]	ADTree, FT, RF	To detect subscriber identity module box (SIMbox) fraud
[114]	LR, SVM-Linear, SVM-RBF, KNN, RF	To predict demographic features such as age and gender
[115]	SVM-Linear, Logistic regression	To predict demographic features such as age and gender
[116]	NB, SVM, DS, RF, RNN	To predict the next location of tourists
[117]	НС	To cluster human mobility patterns based on similar individual trajectories
[118]	GAN	To generate synthetic data of mobile phone data
[119]	KNN, RF, SVM-Linear, SVM-RBF+CNN, LSTM, SDAE	To construct a classifier that enables the recognition of fraudulent phone calls
[120]	RF, GBDT, SVM +CNN	To classify churner customers from non-churner customers
[121]	DT, RF, GBDT, XGBoost	To predict customer churn in Syriatel telecom company
[122]	MLP, SVM, Bayesian networks	To detect prepaid customer churn in mobile telecommunications companies
[123]	RF, DT, MLP, GBDT	To build predictive models that can classify customers into different categories of loyalty, such as very high value customers (greater loyalty), medium value customers (average loyalty), and others
[124]	LDA, SVM-RBF, XGBoost), RF, LR, NB, KNN, Bagged CART, CART, GBDT, C5.0	To predict customer demographic variables such as age and gender in Syriatel Telecom Company
[125]	K-means, DBSCAN	To detect fraudulent calls in telecommunications companies such as

References	Algorithm/Model	Objective
[126]	GMM, ANN	To build a clustering-based classification model to classify cellular network traffic patterns into high-activity area, medium-activity area, low-activity area, etc.
[92,94,127]	K-means, GMM+CNN	To detect anomalous behavior through the identification of anomalous activities of mobile phone subscribers [92], to detect anomalies in a cellular network such as sleeping cells or unusual high call volume in a given region (traffic activity) [94]
[128]	FCM	To classify mobile subscribers based on extracting their calling features into three classes genuine, fraudulent, and suspicious
[129,130]	HC, k-means, FCM, SVM	To detect fraudulent behaviors in telecom companies such as detecting fraudulent calls
[10]	K-means, FCM, spectral clustering, consensus clustering	To cluster land use in Madrid
[131]	FKNN, MLP, C4.5, SVM GBDT, LR, RF, Adaptive boosting	To classify mobile customers into two classes, churners or non-churners
[132]	K-means	To cluster users according to their weekly mobility patterns into six different profiles

Table 7. Cont.

As the table illustrates, most problems in mobile data are regarded as classification and clustering problems; for example, the churn prediction problem has been solved with supervised classification algorithms such as SVM and RF, while the problem of detecting anomalies in mobile phone data has been mostly solved with unsupervised clustering approaches such as k-means clustering.

In addition, supervised algorithms such as RF and SVM are the most widely used in the literature, and at 28 times, this is the highest share among all supervised algorithms. Meanwhile, k-means was the most frequently used unsupervised algorithm, as shown in Figure 3a. On the other hand, deep learning algorithms such as CNN and MLP are the most frequently used for classification problems, including the prediction of customer churn [131].



Figure 3. The distribution of machine learning algorithms (a) and deep learning algorithms (b).

However, because mobile phone data are not labeled, the results above do not reflect that the supervised methods are suitable for other problems, such as land use classification and transport mode estimation, where semi-supervised learning techniques are used to tackle the lack of ground-truth data by relying on small subsets of labeled data, such as in [8,53,94,102].

Finally, some studies combine two or more algorithms to improve the classifier; in this approach, one algorithm works as a preprocessing mechanism to create a user profile, and the second algorithm is used to take these generated user profile clusters as input to the SVM classifier, as in [128,130]. Another example is studies [122,124], which use multiple algorithms, where unsupervised algorithms such as k-means and PCA are used for feature dimensionality reduction and supervised algorithms such as SVM and RF are selected according to which will achieve the best classification results.

6.4. Managerial Implications

This section presents some of the significant managerial implications of using mobile phone data.

Legal and ethical implications: Mobile phone data have benefited our daily lives in many ways, including preventing and controlling infectious diseases and assisting in the fight against criminal activities. However, mobile phone data are subject to privacy breaches because it contains sensitive information about individuals, such as the location of their homes, their most commonly contacted numbers, and their most frequently visited locations. This raises privacy concerns and ethical questions about using such data. Taylor [133] discusses some theoretical and practical implications of using mobile phone data, such as in situations involving a post-conflict state, where the ability to identify a group's location, leaders, and communication networks may give hostile states a way to jeopardize the safety of the people whom they identify.

Future implications: The study's implications revealed that spatiotemporal call volume features have been widely used to investigate the correlations between land use and human activities, to measure ambient population density, and to estimate population densities. They also found that there is not a lot of crime research literature about mobile phone data and that more crime applications, such as finding terrorist networks and credit card fraud, need to be found and studied.

7. Research Opportunities

The current state of mobile phone data in the context of detecting criminal activities and dynamics is still incomplete due to challenges with missing values and partial information (incomplete mobility and calling information). For example, [3,34,35,37] detected criminal behaviors by extracting criminals' call features, such as outgoing and incoming calls, call frequency, maximum and minimum numbers of incoming or outgoing calls and messages, call timestamps, temporal changes in mobile phone call patterns, caller ID, and called ID, to detect criminal communication behaviors. On the other hand, [5,32,42] detected criminal mobility patterns by extracting spatiotemporal features such as variation over time, frequency, and distance. These features allow for the depiction of the mobile phone spatiotemporal patterns of criminals over a given period and allow for the measurement of the distance between the locations where the criminals visited and their homes. However, the results are still limited in providing a more comprehensive picture of all aspects of criminal behavior in terms of mobility patterns, communication behaviors, and social interactions. Consequently, there is a need to improve the detection of criminal behaviors by examining and considering the correlation of these analytical perspectives, which may ultimately provide different aspects of criminal activities.

Anomaly detection in mobile phone data is used to identify anomalies and outliers in the data by detecting anomalous activities (calling and mobility behaviors) or outliers' calls in mobile cellular networks. The literature shows a number of anomaly detection applications that vary based on the characteristics extracted from mobile phone data, such as spatiotemporal features and call characteristics.

Phone fraud detection is one of the primary anomaly applications in mobile phone data. Fraud detection is used to detect fraudulent activities in telecommunications companies. For example, [128] and [130] detected fraudulent calls in mobile telecommunication networks based on extracting mobile subscribers' calling behaviors. To depict subscribers' calling behaviors, the authors extracted calling features from CDRs data, such as type of calls, call duration, frequency of a call, and call timestamp, which were later used to help classify mobile subscribers into three categories: genuine, fraudulent, and suspicious.

Another application of anomaly detection in mobile phone data is detecting emergency events or situations where the data are analyzed to detect population anomalies (abnormal population distribution), such as in [134,135], who estimated the crowd size (number of people in a given location) based on mobile phone usage volume. Here, detecting population anomalies is possible by detecting an abnormal increase in call volumes. Empirically, mobile phone usage or activity that is recorded at given locations based on a base station location where phone calls or SMS volume (high volume of incoming and outgoing phone calls and SMS messages) is much higher than in normal situations could be a sign of a disaster. Therefore, detecting population anomalies can help avoid situations, such as crowd disasters [135].

Another way to detect population anomalies or to detect anomalies in human behaviors was reported by [136], who extracted population trajectories represented by their spatiotemporal features (movement frequency) along with their call behaviors (total outgoing and incoming call volumes, call frequency). These features help depict locations or areas with a higher than normal movement frequency (abnormal movement frequency), which could be a sign of joyous events, such as Christmas Eve or New Year's Eve gatherings. Unusually high call volume and movement frequency could also be a sign of a natural disaster, such as earthquakes or floods.

Other anomaly detection applications have been discussed in the literature, such as detecting anomalies in a cellular network by identifying sleeping cells and traffic overload [92,94,127]. Research on the current state of anomaly detection applications related to mobile phone data is still lacking. Thus, a systematic review is needed to fill in the gaps and investigate the current state of mobile phone data research on anomaly detection by discussing anomaly detection techniques and applications. Furthermore, many applications that are still lacking in the literature, such as malicious activity, credit card fraud detection, cyber intrusion, and terrorist activity, could be investigated.

Churn prediction is also an application that has been discussed widely in mobile phone data. For example, studies [137–139] have applied social network analytics methods (e.g., relational classifier (RC) and community detection algorithm) to predict customer churn, while machine learning techniques have been used in other studies [120–122]. In addition, multiple features have been extracted to depict customers' calling behavior. For example, [120] extracted the number of outgoing calls and SMS, the number of incoming calls, the number of international calls and SMS, the total call duration, the number of incoming/outgoing MMS, and the data uploaded or downloaded volume per subscription. Similarly, [121] extracted the same features with additional features, including the use of the Internet, the average upload/download Internet access, and Internet usage for each customer per day. Therefore, it would also be beneficial to provide a survey that discusses the different features, different applications (such as the prediction for prepaid customers or postpaid subscribers in cellular telecommunication), and models used to predict customer churn.

Many studies have made their dataset privately available (especially mobile phone data at the individual level, known as CDRs data) due to privacy concerns, in which it

contains sensitive details such as spatial-temporal trajectory and communication information about the receiving side of the communication as opposed to mobile phone data at the cell tower level, which does not reveal communication details, while publicly available datasets are mobile phone data aggregated at the cell tower level; thus, we encourage the need for a standard data collection framework in which it provides technical solutions as well as transforming mobile phone data remotely access to third parties by using the privacy-through-security approach, which will be available for interest researchers.

8. Privacy Concerns and Ethical Implications

This section discusses privacy concerns and ethical considerations about the fair use of mobile phone data. Due to the sensitive information that mobile phone data contains, it raises privacy and security concerns about its use for unethical purposes. The risk comes from the breaches of individual significant locations and the leak of their social network communications. Therefore, some technical, practical, and theoretical solutions have been provided to mitigate privacy concerns and ethical challenges.

For example, to mitigate ethical concerns, [140] proposed an ethical framework that defines ethical principles about using mobile phone data. These ethical principles are beneficence, respect for persons, justice, respect for the law, and the public interest. De Montjoye et al. [141] advised establishing an outside ethical committee to evaluate how mobile phone data are used.

However, most researchers have been exploring privacy-preserving techniques and tools that aim to mitigate privacy risks. For example, to maintain privacy and preserve individual and group privacy, [142] suggested applying differential privacy (DP) techniques to mobile phone data. DP protects privacy by injecting a desirable amount of noise into the sensitive data while maintaining a healthy trade-off between privacy and accuracy [143,144]. Pratesi et al. [145] proposed a framework called PRIMULE that can recognize risky user profiles in mobile phone data. The proposed framework aims firstly to mitigate the privacy risk of any set of profiles to which they apply the k-anonymity algorithm, a clustering algorithm that aims to ensure that each individual in a dataset cannot be distinguished from at least k - 1 individuals whose information are also in the dataset. The second aim of PRIMULE is to avoid the misrepresentation of data (data distortion) during the analysis process in order to guarantee that the quality of the profiles is high in terms of similarity with respect to the original ones. Gramaglia et al. [146] proposed anonymization techniques to solve the privacy concerns regarding the disclosure of individual trajectories by applying generalization and suppression methods, one of the anonymity techniques that aim to achieve *k*-anonymity in the spatiotemporal trajectory of mobile phone data. The generalization of spatiotemporal information is carried out by minimizing the precision of trajectory samples in space and time, resulting in indistinguishability between the samples of two or more users. At the same time, suppression eliminates samples that are too difficult to anonymize from the trajectory information in mobile phone data.

Nonetheless, as those approaches are only providing anonymization solutions and are not providing a complete solution to all privacy and security concerns [140] and other issues related to ethics [141], there is still the possibility of re-identifying individuals [133]. Therefore, the privacy and security implications of mobile phone data still represent the most challenging barrier to the growing research effort [147]. Furthermore, without efficient techniques to resolve privacy issues and other ethical considerations, this is not going to change anytime soon [148].

9. Conclusions and Limitations

It is not surprising that mobile phone data have been used as a real-time sensor of human activities and dynamics. This is due to the characteristics of mobile phone data that record users' interactions or activities on a mobile network, such as call and mobility activities, which can show how people move, socialize, and communicate. This in turn explains the explosion of mobile phone data in various disciplines and applications. In this survey, we reviewed many applications driven by mobile phone data, such as urban sensing applications, crime applications, land use classification, anomaly detection, suspect classification, criminal networks, and social network applications. We also discussed several spatiotemporal and mobility features extracted from mobile phone data and differentiated them, explaining their functionality. We also shed light on open opportunities and challenges in mobile phone data. In other words, this study focused on three major subtopics: (1) the current state of mobile phone data applications; (2) techniques and methods used to predict and model human behaviors and mobility patterns; and (3) spatiotemporal and call features and characteristics extracted from mobile phone data to depict human activities and mobility patterns.

The present study aimed to investigate the current state of mobile phone data and its applications and to examine and explore human attributes and characteristics that can be derived from such data. In addition, this survey aimed to shed light on related studies or contributions made in mobile phone data research that the literature lacks, such as anomaly detection, churn prediction, and privacy concerns across various disciplines. Eight electronic databases are used for the retrieval of relevant papers. We filtered the results for articles published between 2013 and 2021 according to our inclusion and exclusion criteria. This allowed us to choose 148 articles for review before we synthesized them.

The results show that the spatiotemporal calling feature has been widely used to depict human behaviors. This feature allows for the estimation and calculation of the number of mobile phone devices or phone calls contacted or managed at a given cell tower at various temporal scales (e.g., hourly, daily, and weekly). This allows for the estimation of hourly dynamic population, the classification of land use, the estimation of ambient population, the investigation of the relationship between human dynamics and crime spatial–temporal patterns, defining the actual populations at risk, and others. The results also show that classification and clustering approaches have been widely used where algorithms such as SVM and k-means are used to classify or cluster human activities based on their calling or mobility features.

Although this study intends to survey mobile phone data methods and techniques in multiple domains, the results are not exhaustive, nor should they be considered a conclusive synthesis of all relevant studies. First, it only surveys contributions made between 2013 and 2021, with a primary focus on the areas of crime and urban research. Second, the study excludes academic content published in languages other than English.

To conclude, opportunities can also be found in mobile phone data, as newly developed methods may enable novel applications such as investigating relationships between human mobility patterns and infectious diseases such as COVID-19. Mobile phone data proved its effectiveness in estimating people's travel and mobility patterns during the spread of infectious diseases and in measuring people's daily mobility exposure to air pollution. Therefore, additional applications can be explored in the future.

Author Contributions: Conceptualization, M.O. and L.Y.P.; methodology, M.O, L.Y.P. and T.F.A.; formal analysis, M.O., L.Y.P., T.F.A. and C.S.K.; writing—original draft preparation, M.O. and L.Y.P.; writing—review and editing, M.O., L.Y.P., T.F.A. and C.S.K.; supervision, L.Y.P., C.S.K. and T.F.A.; project administration, L.Y.P., T.F.A. and C.S.K.; resources, L.Y.P. and T.F.A. All authors have read and agreed to the published version of the manuscript.

Funding: This research was supported by the Malaysian Ministry of Higher Education through the Fundamental Research Grant Scheme (FRGS) grant number FRGS/1/2018/ICT04/UM/02/17.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

Abbreviations

Abbreviations	Definition
COVID-19	Coronavirus disease 2019
CDRs	Call detail records
COVID-19	Coronavirus disease 2019
SNA	Social network analysis
LSOAs	Lower super output areas
SVM	Support vector machines
RF	Random forest
KNN	K-nearest neighbors
FCM	Fuzzy c-means
GCN	Graph convolutional network
GMM	Gaussian mixture model
DBSCAN	Density-based spatial clustering of applications with noise
GBDT	Gradient boosting decision tree
GAN	Generative adversarial network
MLP	Multi-layer perceptron
DT	Decision trees
BP	Backward propagation
HC	Hierarchical clustering
LR	Logistic regression
LSTM	Long short-term memory
PCA	Principal component analysis
SDAE	Stacked denoising autoencoder
XGBoost	Extreme gradient boosting
LDA	Linear discriminant analysis
SVM-Linear	SVM with linear kernel
SVM-RBF	SVM with radial basis function kernel
CART	Classification and regression tree (CART)
Bagged CART	Bagging classification and regression trees
FRNN	Fuzzy-rough nearest neighbors
FKNN	Fuzzy k-nearest neighbor
REDCAP	Regionalization with dynamically constrained agglomerative Clustering and partitioning
SLPA	Speaker–listener label propagation algorithm
SMS	Short message service
ML	Machine learning
DL	Deep learning
PAM	Partitioning around medoids
	0

References

- Blondel, V.D.; Decuyper, A.; Krings, G. A survey of results on mobile phone datasets analysis. *EPJ data science* 2015, 4, 10. [CrossRef]
- Ghahramani, M.; Zhou, M.; Wang, G. Urban sensing based on mobile phone data: Approaches, applications, and challenges. IEEE/CAA J. Autom. Sin. 2020, 7, 627–637. [CrossRef]
- 3. Taha, K.; Yoo, D. SIIMCO: A forensic investigation tool for identifying the influential members of a criminal organization. *IEEE Trans. Inf. Secur.* **2015**, *11*, 811–822. [CrossRef]
- 4. Hassan, S.U.; Shabbir, M.; Iqbal, S.; Said, A.; Kamiran, F.; Nawaz, R.; Saif, U. Leveraging deep learning and SNA approaches for smart city policing in the developing world. *Int. J. Inf. Manag.* **2019**, *56*, 102045. [CrossRef]
- Griffiths, G.; Johnson, S.D.; Chetty, K. UK-based terrorists' antecedent behavior: A spatial and temporal analysis. *Appl. Geogr.* 2017, *86*, 274–282. [CrossRef]
- Eagle, N.; Pentland, A.; Lazer, D. Inferring friendship network structure by using mobile phone data. *Proc. Natl. Acad. Sci. USA* 2009, 106, 15274–15278. [CrossRef]
- Deville, P.; Linard, C.; Martin, S.; Gilbert, M.; Stevens, F.R.; Gaughan, A.E.; Blondel, V.D.; Tatem, A.J. Dynamic population mapping using mobile phone data. *Proc. Natl. Acad. Sci. USA* 2014, 111, 15888–15893. [CrossRef]
- Pei, T.; Sobolevsky, S.; Ratti, C.; Shaw, S.L.; Li, T.; Zhou, C. A new insight into land use classification based on aggregated mobile phone data. *Int. J. Geogr. Inf. Sci.* 2014, 28, 1988–2007. [CrossRef]

- 9. Mao, H.; Ahn, Y.Y.; Bhaduri, B.; Thakur, G. Improving land use inference by factorizing mobile phone call activity matrix. *J. Land Use Sci.* **2017**, *12*, 138–153. [CrossRef]
- 10. Frias–Martinez, V.; Soto, V.; Sánchez, A.; Frias–Martinez, E. 2014. Consensus clustering for urban land use analysis using cell phone network data. *Int. J. Ad Hoc Ubiquitous Comput.* **2014**, *17*, 39–58. [CrossRef]
- 11. Jia, Y.; Ge, Y.; Ling, F.; Guo, X.; Wang, J.; Wang, L.; Chen, Y.; Li, X. Urban land use mapping by combining remote sensing imagery and mobile phone positioning data. *Remote Sens.* **2018**, *10*, 446. [CrossRef]
- Yuan, G.; Chen, Y.; Sun, L.; Lai, J.; Li, T.; Liu, Z. Recognition of functional areas based on call detail records and point of interest data. J. Adv. Transp. 2020, 2020, 8956910. [CrossRef]
- 13. Mao, H.; Thakur, G.; Bhaduri, B. Exploiting mobile phone data for multi-category land use classification in Africa. In Proceedings of the 2nd ACM SIGSPATIAL Workshop on Smart Cities and Urban Analytics, Burlingame, CA, USA, 31 October 2016; pp. 1–6.
- 14. Lenormand, M.; Picornell, M.; Cantú-Ros, O.G.; Louail, T.; Herranz, R.; Barthelemy, M.; Frías-Martínez, E.; San Miguel, M.; Ramasco, J.J. Comparing and modelling land use organization in cities. *R. Soc. Open Sci.* **2015**, *2*, 150449. [CrossRef] [PubMed]
- 15. Ríos, S.A.; Muñoz, R. Land Use detection with cell phone data using topic models: Case Santiago, Chile. *Comput. Environ. Urban Syst.* **2017**, *61*, 39–48. [CrossRef]
- 16. Järv, O.; Tenkanen, H.; Toivonen, T. Enhancing spatial accuracy of mobile phone data using multi-temporal dasymetric interpolation. *Int. J. Geogr. Inf. Sci.* 2017, *31*, 1630–1651. [CrossRef]
- 17. Liu, Z.; Ma, T.; Du, Y.; Pei, T.; Yi, J.; Peng, H. 2018. Mapping hourly dynamics of urban population using trajectories reconstructed from mobile phone records. *Trans. GIS* **2018**, *22*, 494–513. [CrossRef]
- Calabrese, F.; Ferrari, L.; Blondel, V.D. Urban sensing using mobile phone network data: A survey of research. *Acm Comput. Surv. Csur* 2014, 47, 1–20. [CrossRef]
- Bachir, D.; Khodabandelou, G.; Gauthier, V.; El Yacoubi, M.; Puchinger, J. Inferring dynamic origin-destination flows by transport mode using mobile phone data. *Transp. Res. Part C Emerg. Technol.* 2019, 101, 254–275. [CrossRef]
- 20. Jacobs-Crisioni, C.; Rietveld, P.; Koomen, E.; Tranos, E. Evaluating the impact of land-use density and mix on spatiotemporal urban activity patterns: An exploratory study using mobile phone data. *Environ. Plan. A* 2014, *46*, 2769–2785. [CrossRef]
- Dong, Y.; Pinelli, F.; Gkoufas, Y.; Nabi, Z.; Calabrese, F.; Chawla, N.V. Inferring unusual crowd events from mobile phone call detail records. In Proceedings of the Joint European Conference on Machine Learning and Knowledge Discovery in Databases, Porto, Portugal, 7–11 September 2015; Springer: Cham, Germany, 2015; pp. 474–492.
- Furno, A.; El Faouzi, N.E.; Fiore, M.; Stanica, R. Fusing GPS probe and mobile phone data for enhanced land-use detection. In Proceedings of the 2017 5th IEEE International Conference on Models and Technologies for Intelligent Transportation Systems (MT-ITS), Naples, Italy, 26–28 June 2017; IEEE: Piscataway, NJ, USA, 2017; pp. 693–698.
- 23. Zinman, O.; Lerner, B. Utilizing digital traces of mobile phones for understanding social dynamics in urban areas. *Pers. Ubiquitous Comput.* 2020, 24, 535–549. [CrossRef]
- 24. Yang, X.; Fang, Z.; Yin, L.; Li, J.; Lu, S.; Zhao, Z. Revealing the relationship of human convergence–divergence patterns and land use: A case study on Shenzhen City, China. *Cities* **2019**, *95*, 102384. [CrossRef]
- 25. Liu, Y.; Fang, F.; Jing, Y. How urban land use influences commuting flows in Wuhan, Central China: A mobile phone signaling data perspective. *Sustain. Cities Soc.* 2020, *53*, 101914. [CrossRef]
- Novović, O.; Brdar, S.; Mesaroš, M.; Crnojević, V.; Papadopoulos, A.N. Uncovering the relationship between human connectivity dynamics and land use. *ISPRS Int. J. Geo-Inf.* 2020, 9, 140. [CrossRef]
- 27. Long, D.; Liu, L.; Xu, M.; Feng, J.; Chen, J.; He, L. Ambient population and surveillance cameras: The guardianship role in street robbers' crime location choice. *Cities* **2021**, *115*, 103223. [CrossRef]
- Malleson, N.; Andresen, M.A. Exploring the impact of ambient population measures on London crime hotspots. J. Crim. Justice 2016, 46, 52–63. [CrossRef]
- Bogomolov, A.; Lepri, B.; Staiano, J.; Oliver, N.; Pianesi, F.; Pentland, A. Once upon a crime: Towards crime prediction from demographics and mobile data. In Proceedings of the 16th international conference on multimodal interaction, Istanbul, Turkey, 12 November 2014; pp. 427–434.
- 30. Bogomolov, A.; Lepri, B.; Staiano, J.; Letouzé, E.; Oliver, N.; Pianesi, F.; Pentland, A. Moves on the street: Classifying crime hotspots using aggregated anonymized data on people dynamics. *Big Data* **2015**, *3*, 148–158. [CrossRef]
- 31. Rummens, A.; Snaphaan, T.; Van de Weghe, N.; Van den Poel, D.; Pauwels, L.J.; Hardyns, W. Do mobile phone data provide a better denominator in crime rates and improve spatiotemporal predictions of crime? *ISPRS Int. J. Geo-Inf.* **2021**, *10*, 369. [CrossRef]
- 32. Feng, J.; Liu, L.; Long, D.; Liao, W. An examination of spatial differences between migrant and native offenders in committing violent crimes in a large Chinese city. *ISPRS Int. J. Geo-Inf.* **2019**, *8*, 119. [CrossRef]
- 33. He, L.; Páez, A.; Jiao, J.; An, P.; Lu, C.; Mao, W.; Long, D. Ambient population and larceny-theft: A spatial analysis using mobile phone data. *ISPRS Int. J. Geo-Inf.* 2020, *9*, 342. [CrossRef]
- Ferrara, E.; Meo, D.; Catanese, S.; Fiumara, G. Detecting criminal organizations in mobile phone networks. *Expert Syst. Appl.* 2014, 41, 5733–5750. [CrossRef]
- Taha, K.; Yoo, D. Using the spanning tree of a criminal network for identifying its leaders. *IEEE Trans. Inf. Secur.* 2016, 12, 445–453. [CrossRef]
- 36. Taha, K.; Yoo, D. Shortlisting the influential members of criminal organizations and identifying their important communication channels. *IEEE Trans. Inf. Secur.* **2019**, *14*, 1988–1999. [CrossRef]

- Taha, K.; Yoo, D. A system for analyzing criminal social networks. In Proceedings of the 2015 IEEE/ACM International Conference on Advances in Social Networks Analysis and Mining, Paris, France, 25–28 August 2015; pp. 1017–1023.
- Fan, Y.; Yang, T.; Jiang, G.; Zhu, L.; Peng, R. Identifying Criminals' Interactive Behavior and Social Relations through Data Mining on Call Detail Records; DEStech Transactions on Computer Science and Engineering (aiea): Lancaster, PA, USA, 2017.
- 39. Gruber, A.; Ben-Gal, I. Using targeted Bayesian network learning for suspect identification in communication networks. *Int. J. Inf. Secur.* 2018, *17*, 169–181. [CrossRef]
- 40. Dileep, G.K.; Sajeev, G.P. A Graph Mining Approach to Detect Sandwich Calls. In 2021 IEEE International Conference on Electronics, Computing and Communication Technologies (CONECCT); IEEE: Piscataway, NJ, USA, 2021; pp. 1–6.
- Traunmueller, M.; Quattrone, G.; Capra, L. Mining mobile phone data to investigate urban crime theories at scale. In Proceedings of the International Conference on Social Informatics, Barcelona, Spain, 11–13 November 2014; Springer: Cham, Germany, 2014; pp. 396–411.
- 42. Song, G.; Bernasco, W.; Liu, L.; Xiao, L.; Zhou, S.; Liao, W. Crime feeds on legal activities: Daily mobility flows help to explain thieves' target location choices. *J. Quant. Criminol.* **2019**, *35*, 831–854. [CrossRef]
- 43. Hanaoka, K. New insights on relationships between street crimes and ambient population: Use of hourly population data estimated from mobile phone users' locations. *Environ. Plan. B Urban Anal. City Sci.* **2018**, 45, 295–311. [CrossRef]
- 44. Haleem, M.S.; Do Lee, W.; Ellison, M.; Bannister, J. The 'exposed' population, violent crime in public space and the night-time economy in Manchester, UK. *Eur. J. Crim. Policy Res.* **2020**, *27*, 335–352. [CrossRef]
- 45. Lee, W.D.; Haleem, M.S.; Ellison, M.; Bannister, J. The influence of intra-daily activities and settings upon weekday violent crime in public spaces in Manchester, UK. *Eur. J. Crim. Policy Res.* **2020**, *27*, 375–395. [CrossRef]
- 46. Agreste, S.; Catanese, S.; De Meo, P.; Ferrara, E.; Fiumara, G. Network structure and resilience of Mafia syndicates. *Inf. Sci.* 2016, 351, 30–47. [CrossRef]
- 47. Catanese, S.; Ferrara, E.; Fiumara, G. Forensic analysis of phone call networks. Soc. Netw. Anal. Min. 2013, 3, 15–33. [CrossRef]
- 48. Abba, E.; Aibinu, A.M.; Alhassan, J.K. Development of multiple mobile networks call detailed records and its forensic analysis. *Digit. Commun. Netw.* **2019**, *5*, 256–265. [CrossRef]
- Khan, E.S.; Azmi, H.; Ansari, F.; Dhalvelkar, S. Simple implementation of criminal investigation using call data records (CDRs) through big data technology. In Proceedings of the 2018 International Conference on Smart City and Emerging Technology (ICSCET), Mumbai, India, 5 January 2018; IEEE: Piscataway, NJ, USA, 2018; pp. 1–5.
- Kumar, M.; Hanumanthappa, M.; Kumar, T.S. Crime investigation and criminal network analysis using archive call detail records. In Proceedings of the 2016 Eighth International Conference on Advanced Computing (ICoAC), Chennai, India, 19–21 January 2017; IEEE: Piscataway, NJ, USA, 2017; pp. 46–50.
- Khan, S.; Ansari, F.; Dhalvelkar, H.A.; Computer, S. Criminal investigation using call data records (CDR) through big data technology. In Proceedings of the 2017 International Conference on Nascent Technologies in Engineering (ICNTE), Navi Mumbai, India, 27–28 January 2017; IEEE: Piscataway, NJ, USA, 2017; pp. 1–5.
- Tseng, J.C.; Tseng, H.C.; Liu, C.W.; Shih, C.C.; Tseng, K.Y.; Chou, C.Y.; Yu, C.H.; Lu, F.S. A successful application of big data storage techniques implemented to criminal investigation for telecom. In Proceedings of the 2013 15th Asia-Pacific Network Operations and Management Symposium (APNOMS), Hiroshima, Japan, 25–27 September 2013; IEEE: Piscataway, NJ, USA, 2013; pp. 1–3.
- 53. Danya, B. Estimating Urban Mobility with Mobile Network Geolocation Data Mining. PhD Thesis, Université Paris-Saclay, Gif-sur-Yvette, France, 2019.
- 54. Oliver, N.; Matic, A.; Frias-Martinez, E. Mobile network data for public health: Opportunities and challenges. *Front. Public Health* **2015**, *3*, 189. [CrossRef] [PubMed]
- 55. Guo, H.; Li, W.; Yao, F.; Wu, J.; Zhou, X.; Yue, Y.; Yeh, A.G. Who are more exposed to PM2. 5 pollution: A mobile phone data approach. *Environ. Int.* 2020, 143, 105821. [CrossRef] [PubMed]
- Oliver, N.; Lepri, B.; Sterly, H.; Lambiotte, R.; Deletaille, S.; De Nadai, M.; Letouzé, E.; Salah, A.A.; Benjamins, R.; Cattuto, C.; et al. Mobile phone data for informing public health actions across the COVID-19 pandemic life cycle. *Sci. Adv.* 2020, *6*, eabc0764. [CrossRef] [PubMed]
- Grantz, K.H.; Meredith, H.R.; Cummings, D.A.; Metcalf, C.J.E.; Grenfell, B.T.; Giles, J.R.; Mehta, S.; Solomon, S.; Labrique, A.; Kishore, N.; et al. The use of mobile phone data to inform analysis of COVID-19 pandemic epidemiology. *Nat. Commun.* 2020, 11, 4961. [CrossRef] [PubMed]
- Hsiehchen, D.; Espinoza, M.; Slovic, P. Political partisanship and mobility restriction during the COVID-19 pandemic. *Public Health* 2020, *187*, 111–114. [CrossRef]
- 59. Zhou, Y.; Xu, R.; Hu, D.; Yue, Y.; Li, Q.; Xia, J. Effects of human mobility restrictions on the spread of COVID-19 in Shenzhen, China: A modelling study using mobile phone data. *Lancet Digit. Health* **2020**, *2*, e417–e424. [CrossRef]
- 60. Badr, H.S.; Du, H.; Marshall, M.; Dong, E.; Squire, M.M.; Gardner, L.M. Association between mobility patterns and COVID-19 transmission in the USA: A mathematical modelling study. *Lancet Infect. Dis.* **2020**, *20*, 1247–1254. [CrossRef]
- 61. Lai, S.; Farnham, A.; Ruktanonchai, N.W.; Tatem, A.J. Measuring mobility, disease connectivity and individual risk: A review of using mobile phone data and mHealth for travel medicine. *J. Travel Med.* **2019**, *26*, taz019. [CrossRef]
- 62. Dewulf, B.; Neutens, T.; Lefebvre, W.; Seynaeve, G.; Vanpoucke, C.; Beckx, C.; Van de Weghe, N. Dynamic assessment of exposure to air pollution using mobile phone data. *Int. J. Health Geogr.* **2016**, *15*, 14. [CrossRef]

- 63. Yu, X.; Ivey, C.; Huang, Z.; Gurram, S.; Sivaraman, V.; Shen, H.; Eluru, N.; Hasan, S.; Henneman, L.; Shi, G.; et al. Quantifying the impact of daily mobility on errors in air pollution exposure estimation using mobile phone location data. *Environ. Int.* **2020**, 141, 105772. [CrossRef]
- 64. Yin, L.; Lin, N.; Song, X.; Mei, S.; Shaw, S.L.; Fang, Z.; Li, Q.; Li, Y.; Mao, L. Space-time personalized short message service (SMS) for infectious disease control–Policies for precise public health. *Appl. Geogr.* **2020**, *114*, 102103. [CrossRef]
- 65. Chang, H.H.; Wesolowski, A.; Sinha, I.; Jacob, C.G.; Mahmud, A.; Uddin, D.; Zaman, S.I.; Hossain, M.A.; Faiz, M.A.; Ghose, A.; et al. Mapping imported malaria in Bangladesh using parasite genetic and human mobility data. *Elife* 2019, *8*, e43481. [CrossRef] [PubMed]
- 66. Tessema, S.; Wesolowski, A.; Chen, A.; Murphy, M.; Wilheim, J.; Mupiri, A.R.; Ruktanonchai, N.W.; Alegana, V.A.; Tatem, A.J.; Tambo, M.; et al. Using parasite genetic and human mobility data to infer local and cross-border malaria connectivity in Southern Africa. *Elife* **2019**, *8*, e43510. [CrossRef]
- Ihantamalala, F.A.; Herbreteau, V.; Rakotoarimanana, F.M.; Rakotondramanga, J.M.; Cauchemez, S.; Rahoilijaona, B.; Pennober, G.; Buckee, C.O.; Rogier, C.; Metcalf, C.J.E.; et al. Estimating sources and sinks of malaria parasites in Madagascar. *Nat. Commun.* 2018, *9*, 3897. [CrossRef] [PubMed]
- 68. Sekimoto, Y.; Sudo, A.; Kashiyama, T.; Seto, T.; Hayashi, H.; Asahara, A.; Ishizuka, H.; Nishiyama, S. Real-time people movement estimation in large disasters from several kinds of mobile phone data. In Proceedings of the 2016 ACM International Joint Conference on Pervasive and Ubiquitous Computing: Adjunct, Heidelberg, Germany, 12–16 September 2016; pp. 1426–1434.
- 69. Huang, H.; Cheng, Y.; Weibel, R. Transport mode detection based on mobile phone network data: A systematic review. *Transp. Res. Part C Emerg. Technol.* **2019**, *101*, 297–312. [CrossRef]
- Zhong, G.; Yin, T.; Zhang, J.; He, S.; Ran, B. Characteristics analysis for travel behavior of transportation hub passengers using mobile phone data. *Transportation* 2019, 46, 1713–1736. [CrossRef]
- 71. Zhong, G.; Zhang, J.; Li, L.; Chen, X.; Yang, F.; Ran, B. Analyzing passenger travel demand related to the transportation hub inside a city area using mobile phone data. *Transp. Res. Rec.* **2018**, 2672, 23–34. [CrossRef]
- 72. Graells-Garrido, E.; Caro, D.; Parra, D. Inferring modes of transportation using mobile phone data. *EPJ Data Sci.* **2018**, *7*, 49. [CrossRef]
- 73. Chin, K.; Huang, H.; Horn, C.; Kasanicky, I.; Weibel, R. Inferring fine-grained transport modes from mobile phone cellular signaling data. *Comput. Environ. Urban Syst.* 2019, 77, 101348. [CrossRef]
- 74. Lwin, K.K.; Sekimoto, Y. Identification of various transport modes and rail transit behaviors from mobile CDR data: A case of Yangon City. *Asian Transp. Stud.* 2020, *6*, 100025.
- 75. Demissie, M.G.; de Almeida Correia, G.H.; Bento, C. Intelligent road traffic status detection system through cellular networks handover information: An exploratory study. *Transp. Res. Part C Emerg. Technol.* **2013**, *32*, 76–88. [CrossRef]
- 76. Iqbal, M.S.; Choudhury, C.F.; Wang, P.; González, M.C. Development of origin–destination matrices using mobile phone call data. *Transp. Res. Part C Emerg. Technol.* 2014, 40, 63–74. [CrossRef]
- 77. Yang, T. Understanding commuting patterns and changes: Counterfactual analysis in a planning support framework. *Environ. Plan. B Urban Anal. City Sci.* **2020**, *47*, 1440–1455. [CrossRef]
- Steenbruggen, J.; Tranos, E.; Nijkamp, P. Data from mobile phone operators: A tool for smarter cities? *Telecommun. Policy* 2015, 39, 335–346. [CrossRef]
- 79. Phithakkitnukoon, S.; Horanont, T.; Lorenzo, G.D.; Shibasaki, R.; Ratti, C. August. Activity-aware map: Identifying human daily activity pattern using mobile phone data. In *International Workshop on Human Behavior Understanding*; Springer: Berlin, Heidelberg, 2010; pp. 14–25.
- 80. Phithakkitnukoon, S.; Sukhvibul, T.; Demissie, M.; Smoreda, Z.; Natwichai, J.; Bento, C. Inferring social influence in transport mode choice using mobile phone data. *EPJ Data Sci.* 2017, *6*, 11. [CrossRef]
- Qu, Y.; Gong, H.; Wang, P. Transportation mode split with mobile phone data. In Proceedings of the 2015 IEEE 18th International Conference on Intelligent Transportation Systems, Gran Canaria, Spain, 15–18 September 2015; IEEE: Piscataway, NJ, USA, 2015; pp. 285–289.
- 82. Schläpfer, M.; Bettencourt, L.M.; Grauwin, S.; Raschke, M.; Claxton, R.; Smoreda, Z.; West, G.B.; Ratti, C. The scaling of human interactions with city size. *J. R. Soc. Interface* **2014**, *11*, 20130789. [CrossRef] [PubMed]
- 83. Camacho, D.; Panizo-LLedot, Á.; Bello-Orgaz, G.; Gonzalez-Pardo, A.; Cambria, E. The four dimensions of social network analysis: An overview of research methods, applications, and software tools. *Inf. Fusion* **2020**, *63*, 88–120. [CrossRef]
- 84. Doyle, C.; Herga, Z.; Dipple, S.; Szymanski, B.K.; Korniss, G.; Mladenić, D. Predicting complex user behavior from CDR based social networks. *Inf. Sci.* 2019, *500*, 217–228. [CrossRef]
- 85. Gaito, S.; Quadri, C.; Rossi, G.P.; Zignani, M. 2017. Urban communications and social interactions through the lens of mobile phone data. *Online Soc. Netw. Media* 2017, 1, 70–81. [CrossRef]
- 86. Morales, A.J.; Creixell, W.; Borondo, J.; Losada, J.C.; Benito, R.M. Characterizing ethnic interactions from human communication patterns in Ivory Coast. *Netw. Heterog. Media* **2015**, *10*, 87. [CrossRef]
- 87. Liu, X.; Huang, J.; Lai, J.; Zhang, J.; Senousi, A.M.; Zhao, P. Analysis of urban agglomeration structure through spatial network and mobile phone data. *Trans. GIS* **2021**, *25*, 1949–1969. [CrossRef]
- 88. Mao, H.; Shuai, X.; Ahn, Y.Y.; Bollen, J. Quantifying socio-economic indicators in developing countries from mobile phone communication data: Applications to Côte d'Ivoire. *EPJ Data Sci.* **2015**, *4*, 15. [CrossRef]

- Di Clemente, R.; Luengo-Oroz, M.; Travizano, M.; Xu, S.; Vaitla, B.; González, M.C. Sequences of purchases in credit card data reveal lifestyles in urban populations. *Nat. Commun.* 2018, *9*, 3330. [CrossRef] [PubMed]
- 90. Dannemann, T.; Sotomayor-Gómez, B.; Samaniego, H. The time geography of segregation during working hours. *R. Soc. Open Sci.* **2018**, *5*, 180749. [CrossRef]
- 91. Walsh, F.; Pozdnoukhov, A. Spatial structure and dynamics of urban communities. In Proceedings of the 1st Workshop on Pervasive Urban Applications, San Francisco, CA, USA, 12–15 June 2011.
- 92. Sultan, K.; Ali, H.; Zhang, Z. Call detail records driven anomaly detection and traffic prediction in mobile cellular networks. *IEEE Access* 2018, *6*, 41728–41737. [CrossRef]
- Mededovic, E.; Douros, V.G.; Mähönen, P. Node centrality metrics for hotspots analysis in telecom big data. In Proceedings of the IEEE INFOCOM 2019-IEEE Conference on Computer Communications Workshops (INFOCOM WKSHPS), Paris, France, 29 April–2 May 2019; IEEE: Piscataway, NJ, USA, 2019; pp. 417–422.
- 94. Hussain, B.; Du, Q.; Ren, P. Semi-supervised learning based big data-driven anomaly detection in mobile wireless networks. *China Commun.* **2018**, *15*, 41–57. [CrossRef]
- 95. Douglass, R.W.; Meyer, D.A.; Ram, M.; Rideout, D.; Song, D. High resolution population estimates from telecommunications data. *EPJ Data Sci.* **2015**, *4*, 4. [CrossRef]
- 96. Kung, K.S.; Greco, K.; Sobolevsky, S.; Ratti, C. Exploring universal patterns in human home-work commuting from mobile phone data. *PloS One* **2014**, *9*, e96180. [CrossRef]
- 97. Bianchi, F.M.; Rizzi, A.; Sadeghian, A.; Moiso, C. Identifying user habits through data mining on call data records. *Eng. Appl. Artif. Intell.* **2016**, *54*, 49–61. [CrossRef]
- Johnson, P.; Andresen, M.A.; Malleson, N. Cell towers and the ambient population: A spatial analysis of disaggregated property crime. *Eur. J. Crim. Policy Res.* 2021, 27, 313–333. [CrossRef]
- Wang, Y.; Li, J.; Zhao, X.; Feng, G.; Luo, X.R. Using mobile phone data for emergency management: A systematic literature review. *Inf. Syst. Front.* 2020, 22, 1539–1559. [CrossRef] [PubMed]
- Al-Sahaf, H.; Bi, Y.; Chen, Q.; Lensen, A.; Mei, Y.; Sun, Y.; Tran, B.; Xue, B.; Zhang, M. A survey on evolutionary machine learning. J. R. Soc. New Zealand 2019, 49, 205–228. [CrossRef]
- 101. Liu, H.; Lang, B. Machine learning and deep learning methods for intrusion detection systems: A survey. *Appl. Sci.* **2019**, *9*, 4396. [CrossRef]
- 102. Yu, C.; Wang, N.; Yang, L.T.; Yao, D.; Hsu, C.H.; Jin, H. A semi-supervised social relationships inferred model based on mobile phone data. *Future Gener. Comput. Syst.* **2017**, *76*, 458–467. [CrossRef]
- 103. Chen, G.; Hoteit, S.; Viana, A.C.; Fiore, M.; Sarraute, C. Enriching sparse mobility information in call detail records. *Comput. Commun.* **2018**, 122, 44–58. [CrossRef]
- 104. Andrade, T.; Cancela, B.; Gama, J. Discovering locations and habits from human mobility data. *Ann. Telecommun.* **2020**, *75*, 505–521. [CrossRef]
- 105. Zhang, G.; Rui, X.; Poslad, S.; Song, X.; Fan, Y.; Wu, B. A method for the estimation of finely-grained temporal spatial human population density distributions based on cell phone call detail records. *Remote Sens.* **2020**, *12*, 2572. [CrossRef]
- 106. Gabrielli, L.; Furletti, B.; Giannotti, F.; Nanni, M.; Rinzivillo, S. Use of mobile phone data to estimate visitors mobility flows. In Proceedings of the International Conference on Software Engineering and Formal Methods, Berlin, Germany, 26–30 September 2022; Springer: Cham, Germany, 2015; pp. 214–226.
- 107. Gabrielli, L.; Furletti, B.; Trasarti, R.; Giannotti, F.; Pedreschi, D. City users' classification with mobile phone data. In Proceedings of the 2015 IEEE International Conference on Big Data (Big Data), Santa Clara, CA, USA, 29 October–1 November 2015; IEEE: Piscataway, NJ, USA, 2015; pp. 1007–1012.
- 108. Pinter, G.; Mosavi, A.; Felde, I. Artificial intelligence for modeling real estate price using call detail records and hybrid machine learning approach. *Entropy* **2020**, *22*, 1421. [CrossRef]
- 109. Li, M.; Gao, S.; Lu, F.; Zhang, H. Reconstruction of human movement trajectories from large-scale low-frequency mobile phone data. *Comput. Environ. Urban Syst.* 2019, 77, 101346. [CrossRef]
- 110. Cecaj, A.; Lippi, M.; Mamei, M.; Zambonelli, F. Comparing deep learning and statistical methods in forecasting crowd distribution from aggregated mobile phone data. *Appl. Sci.* 2020, *10*, 6580. [CrossRef]
- 111. Wang, G.; Wu, N. A Comparative Study on Contract Recommendation Model: Using Macao Mobile Phone Datasets. *IEEE Access* **2020**, *8*, 39747–39757. [CrossRef]
- 112. Guo, H.; Zhan, Q.; Ho, H.C.; Yao, F.; Zhou, X.; Wu, J.; Li, W. Coupling mobile phone data with machine learning: How misclassification errors in ambient PM2. 5 exposure estimates are produced? *Sci. Total Environ.* 2020, 745, 141034. [CrossRef] [PubMed]
- Murynets, I.; Zabarankin, M.; Jover, R.P.; Panagia, A. Analysis and detection of SIMbox fraud in mobility networks. In Proceedings of the IEEE INFOCOM 2014-IEEE Conference on Computer Communications, Toronto, ON, USA, 27 April–2 May 2014; IEEE: Piscataway, NJ, USA, 2014; pp. 1519–1526.
- 114. Jahani, E.; Sundsøy, P.; Bjelland, J.; Bengtsson, L.; Pentland, A.S.; de Montjoye, Y.A. Improving official statistics in emerging markets using machine learning and mobile phone data. *EPJ Data Sci.* **2017**, *6*, 3. [CrossRef]

- 115. Sarraute, C.; Blanc, P.; Burroni, J. A study of age and gender seen through mobile phone usage patterns in mexico. In Proceedings of the 2014 IEEE/ACM International Conference on Advances in Social Networks Analysis and Mining (ASONAM 2014), Beijing, China, 17–20 August 2014; IEEE: Piscataway, NJ, USA, 2014; pp. 836–843.
- 116. Chen, N.C.; Xie, W.; Welsch, R.E.; Larson, K.; Xie, J. Comprehensive predictions of tourists' next visit location based on call detail records using machine learning and deep learning methods. In Proceedings of the 2017 IEEE International Congress on Big Data (BigData Congress), Honolulu, HI, USA, 25–30 June 2017; IEEE: Piscataway, NJ, USA, 2017; pp. 1–6.
- 117. Zufiria, J.; Pastor-Escuredo, D.; Úbeda-Medina, L.; Hernandez-Medina, M.A.; Barriales-Valbuena, I.; Morales, A.J.; Jacques, D.C.; Nkwambi, W.; Diop, M.B.; Quinn, J.; et al. Identifying seasonal mobility profiles from anonymized and aggregated mobile phone data. Application in food security. *PloS ONE* **2018**, *13*, e0195714. [CrossRef] [PubMed]
- 118. Hughes, B.; Bothe, S.; Farooq, H.; Imran, A. Generative adversarial learning for machine learning empowered self organizing 5G networks. In Proceedings of the 2019 International Conference on Computing, Networking and Communications (ICNC), Honolulu, HI, USA, 18–21 February 2019; IEEE: Piscataway, NJ, USA, 2019; pp. 282–286.
- 119. Xing, J.; Yu, M.; Wang, S.; Zhang, Y.; Ding, Y. Automated Fraudulent Phone Call Recognition through Deep Learning. *Wirel. Commun. Mob. Comput.* **2020**, 2020, 8853468. [CrossRef]
- Chouiekh, A. Deep convolutional neural networks for customer churn prediction analysis. Int. J. Cogn. Inform. Nat. Intell. (IJCINI) 2020, 14, 1–16. [CrossRef]
- 121. Ahmad, A.K.; Jafar, A.; Aljoumaa, K. Customer churn prediction in telecom using machine learning in big data platform. *J. Big Data* **2019**, *6*, 28. [CrossRef]
- 122. Brânduşoiu, I.; Toderean, G.; Beleiu, H. Methods for churn prediction in the pre-paid mobile telecommunications industry. In Proceedings of the 2016 International conference on communications (COMM), Bucharest, Romania, 9–11 June 2016; IEEE: Piscataway, NJ, USA, 2016; pp. 97–100.
- 123. Wassouf, W.N.; Alkhatib, R.; Salloum, K.; Balloul, S. 2020. Predictive analytics using big data for increased customer loyalty: Syriatel Telecom Company case study. *J. Big Data* 2020, 7, 29. [CrossRef]
- 124. Al-Zuabi, I.M.; Jafar, A.; Aljoumaa, K. Predicting customer's gender and age depending on mobile phone data. *J. Big Data* **2019**, *6*, 18. [CrossRef]
- 125. Jabbar, M.A. Fraud Detection Call Detail Record Using Machine Learning in Telecommunications Company. *Adv. Sci. Technol. Eng. Syst. J.* **2020**, *5*, 63–69. [CrossRef]
- 126. Sultan, K.; Ali, H.; Ahmad, A.; Zhang, Z. Call details record analysis: A spatiotemporal exploration toward mobile traffic classification and optimization. *Information* 2019, *10*, 192. [CrossRef]
- 127. Hussain, B.; Du, Q.; Imran, A.; Imran, M.A. Artificial intelligence-powered mobile edge computing-based anomaly detection in cellular networks. *IEEE Trans. Ind. Inform.* 2019, *16*, 4986–4996. [CrossRef]
- 128. Subudhi, S.; Panigrahi, S. A hybrid mobile call fraud detection model using optimized fuzzy C-means clustering and group method of data handling-based network. *Vietnam J. Comput. Sci.* 2018, *5*, 205–217. [CrossRef]
- 129. Hilas, C.S.; Mastorocostas, A.; Rekanos, I.T. Clustering of telecommunications user profiles for fraud detection and security enhancement in large corporate networks: A case study. *Appl. Math. Inf. Sci.* **2015**, *9*, 1709–1718.
- 130. Subudhi, S.; Panigrahi, S. Use of fuzzy clustering and support vector machine for detecting fraud in mobile telecommunication networks. *Int. J. Secur. Netw.* **2016**, *11*, 3–11. [CrossRef]
- 131. Azeem, M.; Usman, M.; Fong, A.C.M. A churn prediction model for prepaid customers in telecom using fuzzy classifiers. *Telecommun. Syst.* **2017**, *66*, 603–614. [CrossRef]
- 132. Thuillier, E.; Moalic, L.; Lamrous, S.; Caminada, A. Clustering weekly patterns of human mobility through mobile phone data. *IEEE Trans. Mob. Comput.* **2017**, *17*, 817–830. [CrossRef]
- Taylor, L. No place to hide? The ethics and analytics of tracking mobility using mobile phone data. *Environ. Plan. D Soc. Space* 2016, 34, 319–336. [CrossRef]
- 134. Liu, S.; Chen, L.; Ni, L.M. Anomaly detection from incomplete data. *ACM Trans. Knowl. Discov. Data* (*TKDD*) **2014**, *9*, 1–22. [CrossRef]
- 135. Botta, F.; Moat, H.S.; Preis, T. Quantifying crowd size with mobile phone and Twitter data. *R. Soc. Open Sci.* 2015, 2, 150162. [CrossRef]
- 136. Dobra, A.; Williams, N.E.; Eagle, N. Spatiotemporal detection of unusual human population behavior using mobile phone data. *PloS ONE* **2015**, *10*, e0120449. [CrossRef] [PubMed]
- 137. Óskarsdóttir, M.; Bravo, C.; Verbeke, W.; Sarraute, C.; Baesens, B.; Vanthienen, J. Social network analytics for churn prediction in telco: Model building, evaluation and network architecture. *Expert Syst. Appl.* **2017**, *85*, 204–220. [CrossRef]
- 138. Kim, K.; Jun, C.H.; Lee, J. Improved churn prediction in telecommunication industry by analyzing a large network. *Expert Syst. Appl.* **2014**, *41*, 6575–6584. [CrossRef]
- Verbeke, W.; Martens, D.; Baesens, B. Social network analysis for customer churn prediction. *Appl. Soft Comput.* 2014, 14, 431–446.
 [CrossRef]
- 140. Letouzé, E.; Vinck, P.; Kammourieh, L. *The Law, Politics and Ethics of Cell Phone Data Analytics*; Data-Pop Alliance: New York, NY, USA, 2015.
- 141. De Montjoye, Y.A.; Gambs, S.; Blondel, V.; Canright, G.; De Cordes, N.; Deletaille, S.; Engø-Monsen, K.; Garcia-Herranz, M.; Kendall, J.; Kerry, C.; et al. On the privacy-conscientious use of mobile phone data. *Sci. Data* **2018**, *5*, 180286. [CrossRef] [PubMed]

- Kishore, N.; Kiang, M.V.; Engø-Monsen, K.; Vembar, N.; Schroeder, A.; Balsari, S.; Buckee, C.O. Measuring mobility to monitor travel and physical distancing interventions: A common framework for mobile phone data analysis. *The Lancet Digital Health* 2020, 2, e622–e628. [CrossRef]
- Jiang, H.; Li, J.; Zhao, P.; Zeng, F.; Xiao, Z.; Iyengar, A. Location privacy-preserving mechanisms in location-based services: A comprehensive survey. ACM Comput. Surv. CSUR 2021, 54, 1–36. [CrossRef]
- 144. Hassan, M.U.; Rehmani, M.H.; Chen, J. Differential privacy techniques for cyber physical systems: A survey. *IEEE Commun. Surv. Tutor.* **2019**, *22*, 746–789. [CrossRef]
- 145. Pratesi, F.; Gabrielli, L.; Cintia, P.; Monreale, A.; Giannotti, F. PRIMULE: Privacy risk mitigation for user profiles. *Data Knowl. Eng.* 2020, 125, 101786. [CrossRef]
- 146. Gramaglia, M.; Fiore, M.; Tarable, A.; Banchs, A. Preserving mobile subscriber privacy in open datasets of spatiotemporal trajectories. In Proceedings of the IEEE INFOCOM 2017-IEEE Conference on Computer Communications, Atlanta, GA, USA, 1–4 May 2017; IEEE: Piscataway, NJ, USA, 2017; pp. 1–9.
- 147. Ghahramani, M.; Zhou, M.; Hon, C.T. 2018. Mobile phone data analysis: A spatial exploration toward hotspot detection. *IEEE Trans. Autom. Sci. Eng.* 2018, *16*, 351–362. [CrossRef]
- 148. Milusheva, S.; Lewin, A.; Gomez, T.B.; Matekenya, D.; Reid, K. Challenges and opportunities in accessing mobile phone data for COVID-19 response in developing countries. In Proceedings of the COMPASS '21: ACM SIGCAS Conference on Computing and Sustainable Societie, online, 28 June–2 July 2021; Data & Policy. p. 3.

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.